

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Dees, Stephane; Pesaran, Mohammad Hashem; Smith, L. Vanessa; Smith, Ron P.

Working Paper Supply, demand and monetary policy shocks in a multicountry New Keynesian model

CESifo Working Paper, No. 3081

Provided in Cooperation with: Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Dees, Stephane; Pesaran, Mohammad Hashem; Smith, L. Vanessa; Smith, Ron P. (2010) : Supply, demand and monetary policy shocks in a multi-country New Keynesian model, CESifo Working Paper, No. 3081, Center for Economic Studies and ifo Institute (CESifo), Munich

This Version is available at: https://hdl.handle.net/10419/39007

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU

Supply, Demand and Monetary Policy Shocks in a Multi-Country New Keynesian Model

Stephane Dees M. Hashem Pesaran L. Vanessa Smith Ron P. Smith

CESIFO WORKING PAPER NO. 3081 CATEGORY 12: EMPIRICAL AND THEORETICAL METHODS JUNE 2010

An electronic version of the paper may be downloaded• from the SSRN website:www.SSRN.com• from the RePEc website:www.RePEc.org• from the CESifo website:www.CESifo-group.org/wp

Supply, Demand and Monetary Policy Shocks in a Multi-Country New Keynesian Model

Abstract

This paper estimates and solves a multi-country version of the standard DSGE New Keynesian (NK) model. The country-specific models include a Phillips curve determining inflation, an IS curve determining output, a Taylor Rule determining interest rates, and a real effective exchange rate equation. The IS equation includes a real exchange rate variable and a country-specific foreign output variable to capture direct inter-country linkages. In accord with the theory all variables are measured as deviations from their steady states, which are estimated as long-horizon forecasts from a reduced-form cointegrating global vector autoregression. The resulting rational expectations model is then estimated for 33 countries on data for 1980Q1-2006Q4, by inequality constrained IV, using lagged and contemporaneous foreign variables as instruments, subject to the restrictions implied by the NK theory. The multi-country DSGE NK model is then solved to provide estimates of identified supply, demand and monetary policy shocks. Following the literature, we assume that the within country supply, demand and monetary policy shocks are orthogonal, though shocks of the same type (e.g. supply shocks in different countries) can be correlated. We discuss estimation of impulse response functions and variance decompositions in such large systems, and present estimates allowing for both direct channels of international transmission through regression coefficients and indirect channels through error spillover effects. Bootstrapped error bands are also provided for the cross country responses of a shock to the US monetary policy.

JEL-Code: C32, E17, F37, F42.

Keywords: global VAR (GVAR), New Keynesian DSGE models, supply shocks, demand shocks, monetary policy shocks.

Stephane Dees	M. Hashem Pesaran
European Central Bank	Cambridge University
L. Vanessa Smith	Kon P. Smith
Cambridge University	Birkbeck College, London

May 2010

For Stephane Dees, the views expressed in this paper do not necessarily reflect the views of the European Central Bank or the Europystem. The other authors are grateful to the European Central Bank for financial support. We benefitted from comments on preliminary versions of the paper presented at the Conference to Honour Professor Adrian Pagan, Sydney, July 13-14, 2009, and at the ECB workshop on "International Linkages and the Macroeconomy: Applications of GVAR Modelling Approaches", December 9, 2009, Frankfurt-am-Main, Germany.

1 Introduction

Business cycle fluctuations transmit both domestically and through the international economy and it is important to determine the extent to which macroeconomic fluctuations result from exogenous national or global shocks to demand, supply or monetary policy. In this paper we provide measures of the effects of such shocks using a multi-country New Keynesian model. While there is a literature using multi-country vector autoregressions (VARs) to model the international transmission of shocks, including Canova and Ciccarelli, 2009 and Dees, di Mauro, Pesaran and Smith (2007, DdPS) who use a global VAR, (GVAR), so far it has proved difficult to use such reduced-form multi-country VARs to examine the effects of structural shocks with clear economic interpretation.

To identify and measure the relative importance of different types of structural shocks much recent literature has used New Keynesian dynamic stochastic general equilibrium, DSGE, models.¹ Within this literature, structural shocks are associated with errors in the log-linearised versions of the first order conditions for households and firms' optimisation problems, with the variables measured as deviations from the steady states. The number and naming of the shocks tend to be model specific. For instance Smets and Wouters (2007) consider a model with seven shocks that they label as technology, risk premium, investment, government spending, wage mark-ups, price mark-ups and monetary policy, though they can be grouped into supply, demand and monetary policy shocks. These models are then used to examine the effect of identified shocks.²

Most of the literature attempting to measure the effect of structural shocks has assumed a closed-economy setting. Carabenciov *et al.* (2008, p.6) who consider developing multi-country models, state that "Large scale DSGE models show promise in this regard, but we are years away from developing empirically based multi-country versions of these models". The open economy contributions have tended to use either models for two economies of comparable size, such as the euro area and the US (as in de Walque *et al.*, 2005, for example), or small open economy, SOE, models where the rest of the world is treated as exogenous. Lubik and Schorfheide (2007), building on the theoretical contributions of Gali and Monacelli (2005), estimate small-scale structural general equilibrium models, similar to the ones estimated below, for Australia, Canada, New Zealand and UK, but, unlike the approach taken in this paper, they treat each of these economies separately, not allowing for the interactions between them.³ Given the questions they were concerned

¹Other approaches to identification of structural shocks have also been considered in the literature. These include the structural VARs proposed by Blanchard and Quah (1989) and identification by sign restrictions on the impulse responses proposed by Uhlig (2005). It is unclear how these approaches can be extended to multi-country models. The structural VAR identification scheme has also been recently criticised by Carlstrom et al. (2009).

 $^{^{2}}$ As a matter of convenience, we refer to this NK model as structural, since under certain standard theoretical assumptions, its parameters can be related to a set of 'deeper' parameters of technology and tastes, but in this paper we do not take a particular position on this interpretation or on the rational expectations and representative agent assumptions on which these models are based: our goal is to generalise the standard DSGE NK model to a multi-country setting.

³The DSGE models considered by Lubik and Schofheide (2007) also lack any backward components and have been criticised by Fukac and Pagan (2010) as being dynamically misspecified.

with, this was not a problem; but when one wishes to measure the effects of structural shocks in a multi-country framework as we do, one needs to allow for the interactions across economies. Compared to a two-country model, an interacting multi-country model raises a range of additional conceptual and technical questions, in particular about the cross-country correlation of shocks and the determination of exchange rates.

This paper answers these conceptual and technical questions within a general framework for the structural analysis of multi-country interactions. The approach is implemented by estimating and solving a relatively large multi-country New Keynesian (MCNK) model, comprising 33 countries on quarterly data over the period 1979Q1-2006Q4. The country-specific models include a Phillips curve representing the aggregate supply equation, an IS curve representing the aggregate demand equation, a Taylor rule for the monetary policy equation, and a reduced-form real effective exchange rate equation. The IS equation includes an exchange rate variable and a country-specific foreign output variable to capture direct inter-country linkages. The US economy is treated differently because the US dollar is used as the numeraire for exchange rates. As a result the US real exchange rate is equal to the inverse of the US price level, and although a Phillips curve determining inflation is estimated for the US, the model is solved in terms of the log US price level, so that country-specific real exchange rates can be determined.

In accord with the theory all variables are measured as deviations from their steady states, which are estimated explicitly as the long-horizon forecasts obtained from a reduced-form cointegrating global vector autoregressive (GVAR) model advanced in Pesaran, Schuermann and Weiner (2004) and further developed in DdPS. The steady states derived under this approach, by taking full account of any stochastic trends and cointegrating relations that might be present in the historical observations, yield deviations that are stationary as required by the theory. The steady states also have the advantage that they are based on long-run economic models that are theoretically consistent with the short-run DSGE model based on the resultant deviations as discussed in Dees, Pesaran, Smith and Smith (2009, DPSS).

The parameters of the structural equations for each country are estimated by the instrumental variable (IV) method subject to inequality restrictions implied by the macroeconomic theory. These include a restriction on the sum of the coefficients of the backward and forward inflation components of the Phillips curve (PC) in order to avoid indeterminate solutions. There has been some concern in the literature as to whether the parameters of such DSGE models are in fact identified, e.g. Canova and Sala (2009). DPSS argue that a multi-country perspective can help identification by using trade-weighted averages of foreign variables as instruments. When there is a large number of countries, these country-specific foreign averages can be treated as weakly exogenous for estimation, even though they are endogenous to the system as a whole. This allows consistent estimates of the equation parameters and thus of the structural errors.

The resulting multi-country rational expectations model is then solved for a unique stable solution. It turns out that theory restrictions together with the restriction on the coefficients of the inflation variables in the PC are sufficient to arrive at a unique stable solution. To our knowledge this is the first time that such a multi-country New Keynesian model under rational expectations has been estimated and solved for a unique stable solution.

The solution is then used to obtain estimates of the supply, demand and monetary policy shocks for all the 33 countries (when applicable). In accordance with the literature, we assume that the within country supply, demand and monetary policy shocks are pair-wise orthogonal, though shocks of the same type (e.g. supply shocks across different countries) can be correlated. We also allow for non-zero correlations between the structural shocks and the reduced form real exchange rate shocks. When the model is solved, the variables in the global economy (all taken to be endogenously determined) can be written as functions of current and past values of the structural shocks, enabling us to calculate structural impulse responses and variance decompositions that allow for the possible correlations of supply, demand and monetary policy shocks across countries. The model allows both for direct channels of international transmission of shocks through contemporaneous effects of foreign variables and indirect channels through error spillover effects.

The rich structure of the multi-country model allows one to address many issues of interest; we shall focus on two different sets of questions. We first examine the impact of a US monetary policy shock on inflation and output deviations in the US and how these effects are then transmitted to the rest of the world, in particular to China, Japan and the euro area economies. This is a natural question given the dominant role of the US in the world economy and the large literature on the effects of US monetary policy shocks. Secondly we examine the effects of global demand and supply shocks (defined as PPP GDP weighted averages of country-specific shocks) on output, inflation and interest rates, distinguishing between direct and indirect channels of transmission of shocks in the global economy. We also investigate the importance of direct channels of transmission by considering a MCNK model without foreign output effects, and examine the importance of using GVAR deviations by estimating an alternative MCNK specification where output deviations are computed using the Hodrick-Prescott filter. The results confirm the importance of allowing for direct channels of international transmission as well as using a cointegrating model for the computation of stationary output, inflation and interest rate deviations. The impulse response results and their bootstrapped bounds are in line with the main predictions of the NK macroeconomic theory and tend to be qualitatively similar across countries. A summary of the main findings is provided in the concluding section.

The rest of this paper is set out as follows: Section 2 describes the structure of the model: the form of the country-specific models, how the countries are linked, and the solution of the multicountry rational expectations model. Section 3 explains the framework for global shock accounting used to calculate the impulse response functions and forecast error variance decompositions, which describe the effects of composite shocks on composite variables. Section 4 considers the issue of estimating the deviations from the steady states. Section 5 presents the parameter estimates and discusses the theory restrictions imposed to ensure that the multi-country NK model has a stable solution and the parameter estimates have the signs predicted by the NK macroeconomic theory. Section 6 examines the effect of various supply, demand and monetary policy shocks. Section 7 considers various extensions and alternative assumptions. Section 8 concludes.

2 The Multi-Country NK model

2.1 Individual equations of the country-specific models

We first describe the individual equations of the country-specific models, then discuss how they are integrated within a multi-country setting and how the resulting rational expectations model is solved. While the framework used is general, the specific model used for illustration is designed to be as close as possible to the standard three equation closed economy New Keynesian models routinely estimated in the literature.⁴ This standard model is augmented to allow for inter-country linkages and estimated for 33 countries subject to a number of *a priori* restrictions from economic theory.

In particular, we consider a multi-country model composed of N + 1 countries, indexed by i, where i = 0, 1, 2, ..., N. The US, i = 0, is treated differently, since the dollar is used as the numeraire currency. The variables for each country are measured as deviations from the steady states, the measurement of which is discussed in Section 4. For country i = 1, 2, ..., N the variables included are inflation deviations, $\tilde{\pi}_{it}$, output deviations, \tilde{y}_{it} , the interest rate deviations, \tilde{r}_{it} and the real effective exchange rate deviations, $\tilde{r}e_{it}$, except for Saudi Arabia where an interest rate variable is not available. The US model includes only the variables: $\tilde{\pi}_{0t}, \tilde{y}_{0t}$, and \tilde{r}_{0t} , since (as it is shown below) the US real exchange rate is proportional to its price level. We also use country-specific foreign variables, which are trade weighted averages of the corresponding variables for other countries. For example the foreign output variable of country i is defined by $\tilde{y}_{it}^* = \sum_{j=0}^N w_{ij} \tilde{y}_{jt}$, where w_{ij} is the trade weight of country j in the total trade (exports plus imports) of country i. By construction $\sum_{j=0}^N w_{ij} = 1, w_{ii} = 0$.

The treatment of exchange rates is central to the construction of a coherent multi-country model and a more detailed discussion of the issues involved is in order. Denote the log nominal exchange rate of country *i* against the US dollar by e_{it} , and the bilateral log exchange rate of country *i* with respect to country *j* by e_{ijt} . It is easily seen that $e_{ijt} = e_{it} - e_{jt}$, and the log real effective exchange rate of country *i* with respect to its trading partners is then given by

$$re_{it} = \sum_{j=0}^{N} w_{ij}(e_{it} - e_{jt}) + \sum_{j=0}^{N} w_{ij}p_{jt} - p_{it},$$

⁴Ireland (2004), for example, notes that 'The development of the forward-looking microfounded New Keynesian model stands, in the eyes of many observers, as one of the past decade's most exciting and significant achievements in macroeconomics.' As examples of this achievement he cites Clarida *et al.* (1999) and Woodford (2003).

where p_{it} is the log general price level in country *i*. Therefore (recalling that $\sum_{j=0}^{N} w_{ij} = 1$)

$$re_{it} = (e_{it} - p_{it}) - \sum_{j=0}^{N} w_{ij}(e_{jt} - p_{jt}),$$

= $ep_{it} - ep_{it}^{*},$ (1)

where $ep_{it}^* = \sum_{j=0}^{N} w_{ij} ep_{jt}$. Deviations from steady states are defined accordingly as $\tilde{r}e_{it} = \tilde{e}p_{it} - \tilde{e}p_{it}^*$.

For the US, $e_{0t} = 0$, and $ep_{0t} = -p_{0t}$, which is determined by the US Phillips curve equation. Specification of a separate exchange rate equation for the US will not be needed. Accordingly, in what follows we shall consider equations for the log real effective exchange rates for countries i = 1, 2, ..., N, and solve for the N + 1 log real exchange rates, \tilde{ep}_{it} , i = 0, 1, 2, ..., N, with the log US real exchange rate deviations being given by $-\tilde{p}_{0t}$. It is important that possible stochastic trends in the log US price level are appropriately taken into account when computing \tilde{p}_{0t} . This can be achieved by first estimating $\tilde{\pi}_{0t}$ and then cumulating the values of $\tilde{\pi}_{0t}$ to obtain \tilde{p}_{0t} up to an arbitrary constant.

The equations in the country-specific models include a standard Phillips curve (PC), derived from the optimising behaviour of monopolistically competitive firms subject to nominal rigidities, which determines inflation deviations $\tilde{\pi}_{it}$, where $\pi_{it} = p_{it} - p_{i,t-1}$. This takes the form

$$\widetilde{\pi}_{it} = \beta_{ib}\widetilde{\pi}_{i,t-1} + \beta_{if}E_{t-1}\left(\widetilde{\pi}_{i,t+1}\right) + \beta_{iy}\widetilde{y}_{it} + \varepsilon_{i,st}, \ i = 0, 1, ..., N,$$
(2)

where $E_{t-1}(\tilde{\pi}_{i,t+1}) = E(\tilde{\pi}_{i,t+1} | \mathfrak{I}_{i,t-1})$. There are no intercepts included in the equations since deviations from steady state values have mean zero by construction. The error term, $\varepsilon_{i,st}$, is interpreted as a supply shock or a shock to the price-cost margin in country *i*. The parameters are non-linear functions of underlying structural parameters. For instance, suppose that there is staggered price setting, with a proportion of firms, $(1 - \theta_i)$, resetting prices in any period, and a proportion θ_i keeping prices unchanged. Of those firms able to adjust prices only a fraction $(1 - \omega_i)$ set prices optimally on the basis of expected marginal costs. A fraction ω_i use a rule of thumb based on lagged inflation. Then for a subjective discount factor, λ_i , we have

$$\begin{split} \beta_{if} &= \lambda_i \theta_i \phi_i^{-1}, \ \beta_{ib} = \omega_i \phi_i^{-1}, \\ \beta_{iy} &= (1 - \omega_i)(1 - \theta_i)(1 - \lambda_i \theta_i) \phi_i^{-1}, \end{split}$$

where $\phi_i = \theta_i + \omega_i [1 - \theta_i (1 - \lambda_i)]$. Notice that there is no reason for these parameters to be the same across countries with very different market institutions and property rights (which will influence λ_i), so we allow them to be heterogeneous from the start. If $\omega_i = 0$, all those who adjust prices do so optimally, then $\beta_{fi} = \lambda_i$, and $\beta_{bi} = 0$. Since $\theta_i \ge 0$, $\omega_i \ge 0$, $\lambda_i \ge 0$ the theory implies $\beta_{ib} \ge 0$, $\beta_{if} \ge 0$, and $\beta_{iy} \ge 0$, which we impose in estimation. The restriction $\beta_{ib} + \beta_{if} < 1$ ensures a unique rational expectations solution in the case where \tilde{y}_{it} is exogenously given and there are no feedbacks from lagged values of inflation to the output gap. The corresponding condition in a multi-country model is likely to be more complicated. We use the restriction $\beta_{ib} + \beta_{if} \leq 0.99$, where the equality corresponds to a 4% per annum discount rate which is often imposed, but this condition might not be sufficient for the model to have a unique solution.

The aggregate demand or IS curve is obtained by log-linearising the Euler equation in consumption and substituting the result in the economy's aggregate resource constraint. In the standard closed economy case, this yields an equation for the output gap, \tilde{y}_{it} , which depends on the expected future output gap, $E_{t-1}(\tilde{y}_{i,t+1})$, and the real interest rate deviations, $\tilde{r}_{it} - E_{t-1}(\tilde{\pi}_{i,t+1})$. Lagged output will enter the IS equation if the utility of consumption for country *i* at time *t* is $u(C_{it} - h_i C_{i,t-1})$ where h_i is a habit persistence parameter. For an open economy model, the aggregate resource constraint will also contain net exports, which in turn will be a function of the real effective exchange rate, \tilde{r}_{it} , and the foreign output gap, \tilde{y}_{it}^* . The open economy version of the standard IS equation is then

$$\widetilde{y}_{it} = \alpha_{ib}\widetilde{y}_{i,t-1} + \alpha_{if}E_{t-1}\left(\widetilde{y}_{i,t+1}\right) + \alpha_{ir}[\widetilde{r}_{it} - E_{t-1}\left(\widetilde{\pi}_{i,t+1}\right)] + \alpha_{ie}\widetilde{r}e_{it} + \alpha_{iy*}\widetilde{y}_{it}^* + \varepsilon_{i,dt}, \ i = 0, 1, \dots, N.$$

The coefficient of the real interest rate, α_{ir} , is interpreted as the inter-temporal elasticity of consumption, see Clarida *et al.* (1999), while $\alpha_{if} = 1/(1 + h_i)$ and $\alpha_{ib} = h_i/(1 + h_i)$. The error, $\varepsilon_{i,dt}$, is interpreted as a demand shock. A number of authors note that unless technology follows a pure random walk process, $\varepsilon_{i,dt}$ may reflect technology shocks, though by conditioning on the foreign output variable the convolution of demand shocks with technology shocks might be somewhat obviated. As discussed further in Section 5, the unrestricted estimates of this equation in the case of many countries resulted in a positive coefficient on the interest rate variable, and given the importance of the interest rate effects in the standard model we decided to impose the restriction $\alpha_{if} = 0$ for all i.⁵ Thus the IS equation used in the model is

$$\widetilde{y}_{it} = \alpha_{ib}\widetilde{y}_{i,t-1} + \alpha_{ir}[\widetilde{r}_{it} - E_{t-1}\left(\widetilde{\pi}_{i,t+1}\right)] + \alpha_{ie}\widetilde{r}e_{it} + \alpha_{iy*}\widetilde{y}_{it}^* + \varepsilon_{i,dt}, \ i = 0, 1, ..., N,$$
(3)

subject to the restrictions $\alpha_{ir} \leq 0$, $\alpha_{iy*} \geq 0$. The analysis of the more general case where $\alpha_{if} \neq 0$, might require consideration of other factors such as financial as well as real variables. But such an extension is beyond the scope of the present paper and will not be pursued here.

The interest rate deviations in country i, \tilde{r}_{it} , (except for Saudi Arabia where interest rate data are not available) are set according to a standard Taylor rule (TR) of the form:

$$\widetilde{r}_{it} = \gamma_{ib}\widetilde{r}_{i,t-1} + \gamma_{i\pi}\widetilde{\pi}_{it} + \gamma_{iy}\widetilde{y}_{it} + \varepsilon_{i,mt}, \ i = 0, 1, ..., N.$$
(4)

The error $\varepsilon_{i,mt}$ is interpreted as a monetary policy shock.

The log real effective exchange rate deviations, $\tilde{r}e_{it}$, are modelled as a stationary first order

 $^{{}^{5}}$ A similar conclusion has also been reached by Fuhrer and Rudebusch (2004). Using maximum likelihood and GMM estimation methods they are unable to find any evidence that rational expectations of future output significantly affect current output in the US.

autoregression,⁶

$$\widetilde{re}_{it} = \rho_i \widetilde{re}_{i,t-1} + \varepsilon_{i,et}, \quad |\rho_i| < 1, \ i = 1, 2, \dots, N.$$
(5)

As noted earlier we do not need a separate exchange rate equation for the US, since the US log real effective exchange rate is given as an *exact* linear combination of the other N log real effective exchange rates.

Putting equations (2) to (5) together for all 33 countries, the total number of variables in the multi-country model is $k = \sum_{i=0}^{N} k_i = 130$, where k_i is the number of variables in country *i*. For the US, with no exchange rate equation, $k_0 = 3$, for Saudi Arabia, with no interest rate equation, $k_{SA} = 3$, for the other 31 countries $k_i = 4$. With 130 endogenous variables the system is already quite large, but can be readily extended to include oil prices, and financial variables such as real equity prices and long term interest rates. The reduced form GVAR model developed in DdPS does include such variables, but these are excluded from the current exercise since the primary aim here is to analyse a multi-country version of the standard New Keynesian model that excludes financial variables.

The parameters of the multi-country model can be estimated consistently for each country separately by instrumental variables (IV) subject to the theory restrictions referred to above. As instruments, following the argument in DPSS, we use an intercept, the lagged values of the country-specific endogenous variables $\tilde{y}_{i,t-1}$, $\tilde{\pi}_{i,t-1}$, $\tilde{r}_{i,t-1}$, $\tilde{r}_{e_{i,t-1}}$, the current values of the country-specific foreign variables \tilde{y}_{it}^* , $\tilde{\pi}_{it}^*$, \tilde{r}_{it}^* , and the log oil price deviation, \tilde{p}_t^o .⁷ The details of the estimation procedure and the estimation results are discussed further in Section 5.

The estimates of the structural parameters can then be used to estimate the country-specific structural shocks, namely the supply, demand and monetary policy shocks as denoted by $\varepsilon_{i,st}, \varepsilon_{i,dt}$ and $\varepsilon_{i,mt}$, respectively, for i = 0, 1, ..., N. As far as the cross correlations of the structural shocks are concerned we follow the literature and assume that these shocks are pair-wise orthogonal within each country, but allow for the shocks of the same type to be correlated across countries. In a multicountry context it does not seem plausible to assume that shocks of the same type are orthogonal across countries. Consider neighbouring economies with similar experiences of supply disruptions, or small economies that are affected by the same supply shocks originating from a dominant economy. As discussed in Chudik and Pesaran (2010), it is possible to deal with such effects explicitly by conditioning the individual country equations on the current and lagged variables of the dominant economy (if any), as well as on the variables of the neighbouring economies. This has been done partly in the specification of the IS equations. But following such a strategy more generally takes us away from the standard New Keynesian model and will not be pursued here. Instead we shall try to deal with such cross-country dependencies through suitably restricted error spillover effects.

⁶Since the model explains the exchange rate and the forward rate (from domestic and foreign interest rates) it implicitly defines the uncovered interest parity risk premium.

⁷Given the importance of oil prices for the determination of steady state inflation and possibly real exchange rates we included an oil price variable in the reduced form GVAR model which is used for the estimation of the steady states.

We also allow the exchange rate shocks, $\varepsilon_{i,et}$, defined by (5), to have non-zero correlations with the other shocks both within and across the countries. This yields the main case we consider: a block diagonal error covariance matrix which is bordered by non-zero covariances between $\varepsilon_{i,et}$ and $(\varepsilon_{i,st}, \varepsilon_{i,dt}, \varepsilon_{i,mt})$, though we shall also consider other more restricted versions of the covariance matrix. There is also an estimation issue: since the dimension of the endogenous variables, k = 130, is larger than the time series dimension, T, an unrestricted (sample) estimate of the variance covariance matrix of the errors is rank deficient and is not guaranteed to be a positive definite matrix. We discuss this further in Section 5.

2.2 Solution of the multi-country RE model

We now consider linking the country-specific models and solving the resultant multi-country RE model. For all countries i = 0, 1, ..., N, let $\tilde{\mathbf{x}}_{it} = (\tilde{\pi}_{it}, \tilde{y}_{it}, \tilde{r}_{it}, \tilde{e}p_{it})'$ with the associated global $(k+1) \times 1$ vector $\tilde{\mathbf{x}}_t = (\tilde{\mathbf{x}}'_{0t}, \tilde{\mathbf{x}}'_{1t}, ..., \tilde{\mathbf{x}}'_{Nt})'$, so that $\tilde{\mathbf{x}}_{0t}$ includes the redundant US real exchange rate variable. This is because although in the US model $\tilde{e}p_{0t} = -\tilde{p}_{0t}$ and $\tilde{\pi}_{0t}$ are related, $\tilde{e}p_{0t}$ is still needed for the construction of $\tilde{e}p_{it}^*$, i = 0, 1, ..., N that enter the IS equations.

In terms of $\tilde{\mathbf{x}}_{it}$ the country-specific models based on equations (2), (3), (4) and (5) can be written as

$$\mathbf{A}_{i0}\widetilde{\mathbf{x}}_{it} = \mathbf{A}_{i1}\widetilde{\mathbf{x}}_{i,t-1} + \mathbf{A}_{i2}E_{t-1}(\widetilde{\mathbf{x}}_{i,t+1}) + \mathbf{A}_{i3}\widetilde{\mathbf{x}}_{it}^* + \mathbf{A}_{i4}\widetilde{\mathbf{x}}_{i,t-1}^* + \boldsymbol{\varepsilon}_{it}, \text{ for } i = 0, 1, ..., N,$$
(6)

where $\tilde{\mathbf{x}}_{it}^* = (\tilde{y}_{it}^*, \tilde{e}\tilde{p}_{it}^*)'$, and as before $\tilde{y}_{it}^* = \sum_{j=0}^N w_{ij}\tilde{y}_{jt}$, and $\tilde{e}\tilde{p}_{it}^* = \sum_{j=0}^N w_{ij}\tilde{e}\tilde{p}_{jt}$. The expectations are taken with respect to a common global information set formed as the union intersection of the individual country information sets, $\mathfrak{I}_{i,t-1}$.

For US, i = 0

$$\mathbf{A}_{00} = \begin{pmatrix} 1 & -\beta_{0y} & 0 & 0 \\ 0 & 1 & -\alpha_{0r} & -\alpha_{0e} \\ -\gamma_{0\pi} & -\gamma_{0y} & 1 & 0 \end{pmatrix}, \quad \mathbf{A}_{01} = \begin{pmatrix} \beta_{0b} & 0 & 0 & 0 \\ 0 & \alpha_{0b} & 0 & 0 \\ 0 & 0 & \gamma_{0b} & 0 \end{pmatrix},$$
$$\mathbf{A}_{02} = \begin{pmatrix} \beta_{0f} & 0 & 0 & 0 \\ -\alpha_{0r} & 0 & 0 & 0 \\ 0 & 0 & 0 & 0 \end{pmatrix}, \quad \mathbf{A}_{03} = \begin{pmatrix} 0 & 0 \\ \alpha_{0y*} & -\alpha_{0e} \\ 0 & 0 \end{pmatrix},$$

and $\varepsilon_{0t} = (\varepsilon_{0,st}, \varepsilon_{0,dt}, \varepsilon_{0,mt})'$. Note that $\mathbf{A}_{04} = \mathbf{0}$, since there is no exchange rate equation for the US.

For the other countries, i = 1, 2, ..., N, (except Saudi Arabia where there is no interest rate

equation)

and $\boldsymbol{\varepsilon}_{it} = (\varepsilon_{i,st}, \varepsilon_{i,dt}, \varepsilon_{i,mt}, \varepsilon_{i,et})'$.

Let $\widetilde{\mathbf{z}}_{it} = (\widetilde{\mathbf{x}}'_{it}, \widetilde{\mathbf{x}}^{*\prime}_{it})'$ then the N + 1 models specified by (6) can be written compactly as

$$\mathbf{A}_{iz0}\widetilde{\mathbf{z}}_{it} = \mathbf{A}_{iz1}\widetilde{\mathbf{z}}_{i,t-1} + \mathbf{A}_{iz2}E_{t-1}\left(\widetilde{\mathbf{z}}_{i,t+1}\right) + \boldsymbol{\varepsilon}_{it}, \text{ for } i = 0, 1, 2, ..., N,$$
(7)

where

$$\begin{aligned} \mathbf{A}_{0z0} &= (\mathbf{A}_{00}, \ -\mathbf{A}_{03}), \ \mathbf{A}_{0z1} = \left(\mathbf{A}_{01}, \ \mathbf{0}_{k_0 \times (k_0 + 1 + k_0^*)}\right), \ \mathbf{A}_{0z2} = \left(\mathbf{A}_{02}, \ \mathbf{0}_{k_0 \times (k_0 + 1 + k_0^*)}\right), \ \text{for } i = 0, \\ \mathbf{A}_{iz0} &= (\mathbf{A}_{i0}, \ -\mathbf{A}_{i3}), \ \mathbf{A}_{iz1} = \left(\mathbf{A}_{i1}, \ \mathbf{A}_{i4}\right), \ \mathbf{A}_{iz2} = \left(\mathbf{A}_{i2}, \ \mathbf{0}_{k_i \times (k_i + k_i^*)}\right), \ \text{for } i = 1, 2, ..., N. \end{aligned}$$

The variables $\widetilde{\mathbf{z}}_{it}$ are linked to the variables in the global model, $\widetilde{\mathbf{x}}_t$, through the identity

$$\widetilde{\mathbf{z}}_{it} = \mathbf{W}_i \widetilde{\mathbf{x}}_t,\tag{8}$$

where the 'link' matrices \mathbf{W}_i , i = 0, 1, ..., N are defined in terms of the weights w_{ij} . For i = 0, \mathbf{W}_0 is $(k_0 + 1 + k_0^*) \times (k + 1)$ and for i = 1, 2, ..., N, \mathbf{W}_i is $(k_i + k_i^*) \times (k + 1)$ dimensional.

Substituting (8) in (7) now yields

$$\mathbf{A}_{iz0}\mathbf{W}_{i}\widetilde{\mathbf{x}}_{t} = \mathbf{A}_{iz1}\mathbf{W}_{i}\widetilde{\mathbf{x}}_{t-1} + \mathbf{A}_{iz2}\mathbf{W}_{i}E_{t-1}\left(\widetilde{\mathbf{x}}_{t+1}\right) + \boldsymbol{\varepsilon}_{it}, \ i = 0, 1, ..., N,$$

and then stacking all the N + 1 country models we obtain the multi-country RE model for $\tilde{\mathbf{x}}_t$ as

$$\mathbf{A}_0 \widetilde{\mathbf{x}}_t = \mathbf{A}_1 \widetilde{\mathbf{x}}_{t-1} + \mathbf{A}_2 E_{t-1} \left(\widetilde{\mathbf{x}}_{t+1} \right) + \boldsymbol{\varepsilon}_t, \tag{9}$$

where the stacked $k \times (k+1)$ matrices \mathbf{A}_j , j = 0, 1, 2 are defined by

$$\mathbf{A}_{j} = \begin{pmatrix} \mathbf{A}_{0zj} \mathbf{W}_{0} \\ \mathbf{A}_{1zj} \mathbf{W}_{1} \\ \vdots \\ \mathbf{A}_{Nzj} \mathbf{W}_{N} \end{pmatrix}, \ j = 0, 1, 2, \text{ and } \boldsymbol{\varepsilon}_{t} = \begin{pmatrix} \boldsymbol{\varepsilon}_{0t} \\ \boldsymbol{\varepsilon}_{1t} \\ \vdots \\ \boldsymbol{\varepsilon}_{Nt} \end{pmatrix}.$$

The multi-country RE model given by (9) represents a system of k variables in k + 1 RE equations, and as noted above, contains a redundant equation in the US model. To deal with this

redundancy we consider the new $k \times 1$ vector $\tilde{\mathbf{x}}_{t} = (\tilde{\mathbf{x}}'_{0t}, \tilde{\mathbf{x}}'_{1t}, ..., \tilde{\mathbf{x}}'_{Nt})'$, where $\tilde{\mathbf{x}}_{0t} = (y_{0t}, r_{0t}, ep_{0t})'$ and $\tilde{\mathbf{x}}_{it} = \tilde{\mathbf{x}}_{it}$ for i = 1, 2, ..., N. In particular, for the US we can relate the 4×1 vector $\tilde{\mathbf{x}}_{0t}$ to the 3×1 vector $\tilde{\mathbf{x}}_{0t}$ by ,

$$\widetilde{\mathbf{x}}_{0t} = \mathbf{S}_{00} \mathbf{\mathring{x}}_{0t} - \mathbf{S}_{01} \mathbf{\mathring{x}}_{0,t-1},$$

where

$$\mathbf{S}_{00} = \begin{pmatrix} 0 & 0 & -1 \\ 1 & 0 & 0 \\ 0 & 1 & 0 \\ 0 & 0 & 1 \end{pmatrix}, \ \mathbf{S}_{01} = \begin{pmatrix} 0 & 0 & -1 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \\ 0 & 0 & 0 \end{pmatrix}.$$

Similarly, $\tilde{\mathbf{x}}_t = (\tilde{\mathbf{x}}'_{0t}, \tilde{\mathbf{x}}'_{1t}, ..., \tilde{\mathbf{x}}'_{Nt})'$ can be related to the $k \times 1$ global vector $\tilde{\mathbf{x}}_t = (\tilde{\mathbf{x}}'_{0t}, \tilde{\mathbf{x}}'_{1t}, ..., \tilde{\mathbf{x}}'_{Nt})'$ by

$$\widetilde{\mathbf{x}}_t = \mathbf{S}_0 \widetilde{\widetilde{\mathbf{x}}}_t - \mathbf{S}_1 \widetilde{\widetilde{\mathbf{x}}}_{t-1},\tag{10}$$

where

$$\mathbf{S}_{0} = \begin{pmatrix} \mathbf{S}_{00} & \mathbf{0}_{4 \times (k-3)} \\ \mathbf{0}_{(k-3) \times 3} & \mathbf{I}_{k-3} \end{pmatrix}, \mathbf{S}_{1} = \begin{pmatrix} \mathbf{S}_{01} & \mathbf{0}_{4 \times (k-3)} \\ \mathbf{0}_{(k-3) \times 3} & \mathbf{0}_{(k-3) \times (k-3)} \end{pmatrix}$$

Using (10) in (9) we have

$$\mathbf{A}_0\left(\mathbf{S}_0\tilde{\mathbf{x}}_t - \mathbf{S}_1\tilde{\mathbf{x}}_{t-1}\right) = \mathbf{A}_1\left(\mathbf{S}_0\tilde{\mathbf{x}}_{t-1} - \mathbf{S}_1\tilde{\mathbf{x}}_{t-2}\right) + \mathbf{A}_2E_{t-1}(\mathbf{S}_0\tilde{\mathbf{x}}_{t+1} - \mathbf{S}_1\tilde{\mathbf{x}}_t) + \boldsymbol{\varepsilon}_t,$$

or

$$\mathbf{H}_{0}\widetilde{\mathbf{x}}_{t} = \mathbf{H}_{1}\widetilde{\mathbf{x}}_{t-1} + \mathbf{H}_{2}\widetilde{\mathbf{x}}_{t-2} + \mathbf{H}_{3}E_{t-1}(\widetilde{\mathbf{x}}_{t+1}) + \mathbf{H}_{4}E_{t-1}(\widetilde{\mathbf{x}}_{t}) + \boldsymbol{\varepsilon}_{t},$$
(11)

where

$$H_0 = A_0 S_0, H_1 = A_1 S_0 + A_0 S_1, H_2 = -A_1 S_1, H_3 = A_2 S_0, H_4 = -A_2 S_1.$$

For a determinate solution the $k \times k$ matrix \mathbf{H}_0 must be non-singular. Pre-multiplying (11) by \mathbf{H}_0^{-1}

$$\widetilde{\mathbf{x}}_{t} = \mathbf{F}_{1}\widetilde{\mathbf{x}}_{t-1} + \mathbf{F}_{2}\widetilde{\mathbf{x}}_{t-2} + \mathbf{F}_{3}E_{t-1}(\widetilde{\mathbf{x}}_{t+1}) + \mathbf{F}_{4}E_{t-1}(\widetilde{\mathbf{x}}_{t}) + \mathbf{u}_{t},$$
(12)

where $\mathbf{F}_j = \mathbf{H}_0^{-1} \mathbf{H}_j$, for j = 1, 2, 3, 4, and $\mathbf{u}_t = \mathbf{H}_0^{-1} \boldsymbol{\varepsilon}_t$. Using a companion form representation (12) can be written as

$$\boldsymbol{\chi}_t = \mathbf{A}\boldsymbol{\chi}_{t-1} + \mathbf{B}\boldsymbol{E}_{t-1}(\boldsymbol{\chi}_{t+1}) + \boldsymbol{\eta}_t, \tag{13}$$

where $\boldsymbol{\chi}_t = \left(\widetilde{\mathbf{x}}_t', \widetilde{\mathbf{x}}_{t-1}' \right)'$, and

$$\mathbf{A}=\left(egin{array}{cc} \mathbf{F}_1 & \mathbf{F}_2 \ \mathbf{I}_k & \mathbf{0} \end{array}
ight), \ \mathbf{B}=\left(egin{array}{cc} \mathbf{F}_3 & \mathbf{F}_4 \ \mathbf{0} & \mathbf{0} \end{array}
ight), \ oldsymbol{\eta}_t=\left(egin{array}{cc} \mathbf{u}_t \ \mathbf{0} \end{array}
ight).$$

The system of equations in (13) is the canonical rational expectations model and its solution has been considered in the literature. Binder and Pesaran (1995, 1997) review the alternative solution strategies and show that the nature of the solution critically depends on the roots of the quadratic matrix equation

$$\mathbf{B}\boldsymbol{\Phi}^2 - \boldsymbol{\Phi} + \mathbf{A} = \mathbf{0}.$$
 (14)

There will be a unique globally consistent stationary solution if (14) has a real matrix solution such that all the eigenvalues of $\mathbf{\Phi}$ and $(\mathbf{I} - \mathbf{B}\mathbf{\Phi})^{-1}\mathbf{B}$ lie strictly inside the unit circle. The solution is then given by

$$\boldsymbol{\chi}_t = \boldsymbol{\Phi} \boldsymbol{\chi}_{t-1} + \boldsymbol{\eta}_t. \tag{15}$$

Partitioning Φ conformably to χ_t , (15) can be expressed as

$$egin{pmatrix} \widetilde{\mathbf{x}}_t \ \widetilde{\mathbf{x}}_{t-1} \ \end{pmatrix} = egin{pmatrix} \mathbf{\Phi}_{11} & \mathbf{\Phi}_{12} \ \mathbf{I}_k & \mathbf{0} \ \end{pmatrix} egin{pmatrix} \widetilde{\mathbf{x}}_{t-1} \ \widetilde{\mathbf{x}}_{t-2} \ \end{pmatrix} + egin{pmatrix} \mathbf{I}_k & \mathbf{0} \ \mathbf{0} & \mathbf{I}_k \ \end{pmatrix} egin{pmatrix} \mathbf{u}_t \ \mathbf{0} \ \end{pmatrix},$$

so that the solution in terms of $\mathbf{\dot{x}}_t$, is given by

$$\widetilde{\mathbf{x}}_{t} = \mathbf{\Phi}_{11} \widetilde{\mathbf{x}}_{t-1} + \mathbf{\Phi}_{12} \widetilde{\mathbf{x}}_{t-2} + \mathbf{H}_{0}^{-1} \boldsymbol{\varepsilon}_{t}, \qquad (16)$$

where $\boldsymbol{\varepsilon}_t = (\boldsymbol{\varepsilon}_{0t}', \boldsymbol{\varepsilon}_{1t}', ..., \boldsymbol{\varepsilon}_{Nt}')'$. The structural shocks, $\boldsymbol{\varepsilon}_t$, can be recovered by noting that

$$\boldsymbol{\varepsilon}_t = \mathbf{H}_0(\widetilde{\mathbf{x}}_t - \boldsymbol{\Phi}_{11}\widetilde{\mathbf{x}}_{t-1} - \boldsymbol{\Phi}_{12}\widetilde{\mathbf{x}}_{t-2}).$$
(17)

The covariance matrix of the structural shocks is given by

$$E(\boldsymbol{\varepsilon}_t \boldsymbol{\varepsilon}_t') = \boldsymbol{\Sigma}_{\boldsymbol{\varepsilon}},\tag{18}$$

which can be obtained from the estimated structural shocks.

It will be convenient to reorder the elements of ε_t in (17) in terms of the different types of shocks as $\varepsilon_t^0 = (\varepsilon'_{st}, \varepsilon'_{dt}, \varepsilon'_{mt}, \varepsilon'_{et})'$, where ε_{st} and ε_{dt} are the $(N + 1) \times 1$ vectors of supply and demand shocks, and ε_{mt} and ε_{et} are the $N \times 1$ vectors of monetary policy shocks (for all countries except Saudi Arabia) and shocks to the real effective exchange rates (for all countries except the US). We can then write

$$\boldsymbol{\varepsilon}_t^0 = \mathbf{G}\boldsymbol{\varepsilon}_t,\tag{19}$$

where **G** is a non-singular $k \times k$ matrix with elements 0 or 1. Also $E(\varepsilon_t^0 \varepsilon_t^{0'}) = \Sigma_{\varepsilon}^0 = \mathbf{G} \Sigma_{\varepsilon} \mathbf{G}'$, which can be obtained from Σ_{ε} by suitable permutations of its rows and columns.

As discussed above, we assume that there are zero covariances between the supply, demand and monetary policy shocks, though there can be non-zero covariances between the same type of structural shocks in different countries. We allow the exchange rate shocks, ε_{et} , to interact with the other shocks both within and across the countries. The covariance of demand and supply shocks are given by the $(N+1) \times (N+1)$ dimensional matrices Σ_{ss} and Σ_{dd} , and the covariance matrices of the monetary policy shocks and exchange rate shocks are given by the $N \times N$ matrices Σ_{mm} and Σ_{ee} . The covariances between the exchange rate shocks and the structural shocks are given by Σ_{es} , etc. These assumptions yield a block diagonal error covariance matrix which is bordered by non-zero covariances between the exchange rate shocks and the structural shocks, so that Σ_{ε}^{0} has the form:

$$\Sigma_{\varepsilon}^{0} = \begin{pmatrix} \Sigma_{ss} & 0 & 0 & \Sigma_{se} \\ 0 & \Sigma_{dd} & 0 & \Sigma_{de} \\ 0 & 0 & \Sigma_{mm} & \Sigma_{me} \\ \Sigma_{es} & \Sigma_{ed} & \Sigma_{em} & \Sigma_{ee} \end{pmatrix}.$$
(20)

3 Impulse responses, variance decompositions and shock accounting in the MCNK model

The analysis of the effects of shocks will be represented, as usual, by impulse response functions, IRFs, and forecast error variance decompositions, FEVDs. The system is solved in terms of the $k \times 1$ vector $\tilde{\mathbf{x}}_t$, and since $\tilde{\mathbf{x}}_{0t} = (\tilde{y}_{0t}, \tilde{r}_{0t}, \tilde{e}\tilde{p}_{0t})' = (\tilde{y}_{0t}, \tilde{r}_{0t}, -\tilde{p}_{0t})'$, it includes the US price level and not $\tilde{\pi}_{0t}$. To compute the effects of shocks on US inflation we can switch back to the $(k + 1) \times 1$ vector $\tilde{\mathbf{x}}_t$, as defined by (10). The standard approach to IRFs and FEVDs needs to be somewhat modified to deal with the cross-country error correlations and below we discuss their calculation.

3.1 Impulse response functions

Impulse response functions provide counter-factual answers to questions concerning either the effects of a particular shock in a given economy, or the effects of a combined shock involving linear combinations of shocks across two or more economies. The effects of the shock can also be computed either on a particular variable in the global economy or on a combination of variables. Denote a composite shock, defined as a linear combination of the shocks, by $\xi_t = \mathbf{a}' \boldsymbol{\varepsilon}_t^0$, and consider the time profile of its effects on a composite variable $q_t = \mathbf{b}' \tilde{\mathbf{x}}_t$. The $k \times 1$ vector \mathbf{a} and the $(k+1) \times 1$ vector \mathbf{b} are either appropriate selection vectors picking out a particular error or variable or a suitable weighted average. The error weights, \mathbf{a} , can be chosen to define composite shocks, such as a global supply shock; the variable weighted average of the countries in the euro area. The IRFs estimate the time profile of the response by $q_t = \mathbf{b}' \tilde{\mathbf{x}}_t$ to a unit shock (defined as one standard error shock of size $\sigma_{\xi} = \sqrt{\mathbf{a}' \boldsymbol{\Sigma}_{\varepsilon}^0 \mathbf{a}}$) to $\xi_t = \mathbf{a}' \boldsymbol{\varepsilon}_t^0$, and the FEVDs estimate the relative importance of different shocks in explaining the variations in output, inflation and interest rates from their steady states in a particular economy over time.

Using (16) and (19), we obtain

$$\widetilde{\mathbf{x}}_{t} = \mathbf{\Phi}_{11} \widetilde{\mathbf{x}}_{t-1} + \mathbf{\Phi}_{12} \widetilde{\mathbf{x}}_{t-2} + \mathbf{H}_{0}^{-1} \mathbf{G}^{-1} \boldsymbol{\varepsilon}_{t}^{0}, \qquad (21)$$

and the time profile of $\mathbf{\dot{x}}_{t+n}$ in terms of current and lagged shocks can be written as

$$\widetilde{\mathbf{x}}_{t+n} = \mathbf{D}_{n1}\widetilde{\mathbf{x}}_{t-1} + \mathbf{D}_{n2}\widetilde{\mathbf{x}}_{t-2} + \mathbf{C}_n \boldsymbol{\varepsilon}_t^0 + \mathbf{C}_{n-1}\boldsymbol{\varepsilon}_{t+1}^0 + \dots + \mathbf{C}_1 \boldsymbol{\varepsilon}_{t+n-1}^0 + \mathbf{C}_0 \boldsymbol{\varepsilon}_{t+n}^0,$$
(22)

where \mathbf{D}_{n1} and \mathbf{D}_{n2} are functions of Φ_{11} and Φ_{12} , $\mathbf{C}_j = \mathbf{P}_j \mathbf{H}_0^{-1} \mathbf{G}^{-1}$, and \mathbf{P}_j can be derived recursively as

$$\mathbf{P}_{j} = \mathbf{\Phi}_{11}\mathbf{P}_{j-1} + \mathbf{\Phi}_{12}\mathbf{P}_{j-2}, \ \mathbf{P}_{0} = \mathbf{I}_{k}, \ \mathbf{P}_{j} = \mathbf{0}, \ \text{for } j < 0$$

Similarly, using (10) and (21), we have

$$\widetilde{\mathbf{x}}_{t+n} = \mathbf{\mathring{D}}_{n1}\widetilde{\mathbf{\mathring{x}}}_{t-1} + \mathbf{\mathring{D}}_{n2}\widetilde{\mathbf{\mathring{x}}}_{t-2} + \mathbf{B}_n \boldsymbol{\varepsilon}_t^0 + \mathbf{B}_{n-1}\boldsymbol{\varepsilon}_{t+1}^0 + \dots + \mathbf{B}_1 \boldsymbol{\varepsilon}_{t+n-1}^0 + \mathbf{B}_0 \boldsymbol{\varepsilon}_{t+n}^0,$$
(23)

where

$$\mathbf{\check{D}}_{n1} = \mathbf{S}_0 \mathbf{D}_{n1} - \mathbf{S}_1 \mathbf{D}_{n-1,1}, \quad \mathbf{\check{D}}_{n2} = \mathbf{S}_0 \mathbf{D}_{n2} - \mathbf{S}_1 \mathbf{D}_{n-1,2},$$

and

$$\mathbf{B}_0 = \mathbf{S}_0 \mathbf{C}_0$$
 and $\mathbf{B}_\ell = \mathbf{S}_0 \mathbf{C}_\ell - \mathbf{S}_1 \mathbf{C}_{\ell-1}$, for $\ell = 1, 2, ..., n$

Notice that B_{ℓ} , $\ell = 0, 1, 2, ..., n$ are $(k + 1) \times k$, dimensional matrices that transmit the effects of the k shocks in the system to the (k + 1) elements of $\tilde{\mathbf{x}}_{t+n}$ that include both the US price level and the US inflation. Clearly, both representations (22) and (23) can be used to carry out the impulse response analysis. But it is more convenient to use (23) when considering the effects of shocks on US inflation.

The generalized impulse response function for the effect on $q_t = \mathbf{b}' \mathbf{\tilde{x}}_t$ of a one standard error shock to $\xi_t = \mathbf{a}' \boldsymbol{\varepsilon}_t^0$ is then

$$g_{q}(n,\sigma_{\xi}) = E(q_{t+n} \mid \xi_{t} = \sigma_{\xi} = \sqrt{\mathbf{a}' \boldsymbol{\Sigma}_{\varepsilon}^{0} \mathbf{a}}, \mathfrak{I}_{t-1}) - E(\mathbf{b}' \widetilde{\mathbf{x}}_{t+n} \mid \mathfrak{I}_{t-1})$$

$$= \frac{\mathbf{b}' \mathbf{B}_{n} \boldsymbol{\Sigma}_{\varepsilon}^{0} \mathbf{a}}{\sqrt{\mathbf{a}' \boldsymbol{\Sigma}_{\varepsilon}^{0} \mathbf{a}}}, \ n = 0, 1, 2, \dots .$$

$$(24)$$

While we can identify the IRFs of, say, supply shocks as a group because they are assumed to be orthogonal to demand and monetary policy shocks, we cannot identify the supply shock in any particular country, because they are correlated with the supply shocks in other countries. The issue of how to identify country-specific demand or supply shocks in a multi-country setting is beyond the scope of the present paper. Instead here we focus on the effects of global supply or demand shocks. For instance, a global supply shock uses \mathbf{a}_s , which has PPP GDP weights that add to one, corresponding to the supply shocks of each of the N + 1 countries and zeros elsewhere. For a monetary policy shock, we consider a unit (one standard error) shock to the US interest rate and examine its effects on the US and the rest of the world.⁸ We interpret this as the effect of a shock to US monetary policy, which can be justified, for example, in the context of a recursive specification of monetary policy shocks where in the block of interest rate equations the US monetary policy rule is placed first.

3.2 Forecast error variance decomposition

Forecast error variance decomposition (FEVD) techniques can also be used to estimate the relative importance of different types of shocks in explaining the forecast error variance of different variables in the world economy. Such a decomposition can be achieved without having to specify the nature or sources of the cross-country correlations of supply or demand shocks. Additional identifying assumptions will be needed if we also wish to identify the relative importance of country-specific supply shocks, but as noted above such an exercise is beyond the scope of the present paper.

 $^{^{8}}$ Similar issues have been considered by Eichenbaum and Evans (1995) and Kim (2001), but using two-country VARs.

For the FEVD of global shocks we partition $\mathbf{B}_{\ell} = \begin{pmatrix} \mathbf{B}_{s\ell}, \mathbf{B}_{d\ell}, \mathbf{B}_{m\ell}, \mathbf{B}_{e\ell} \end{pmatrix}$ in (23) conformably with the partitioning of $\boldsymbol{\varepsilon}_t^0 = (\boldsymbol{\varepsilon}'_{st}, \boldsymbol{\varepsilon}'_{dt}, \boldsymbol{\varepsilon}'_{mt}, \boldsymbol{\varepsilon}'_{et})'$, and note that the *n* step ahead forecast errors can be written as

$$\tilde{\boldsymbol{v}}_{t+n} = \widetilde{\mathbf{x}}_{t+n} - E\left(\widetilde{\mathbf{x}}_{t+n} \left| \mathfrak{I}_{t-1} \right.\right) = \sum_{j \in shocks} \sum_{\ell=0}^{n} \mathbf{B}_{j,n-\ell} \boldsymbol{\varepsilon}_{j,t+\ell}.$$
(25)

Under the assumption that within country supply, demand and monetary policy shocks are orthogonal we have

$$Var\left(\tilde{\boldsymbol{\upsilon}}_{t+n} | \boldsymbol{\Im}_{t-1}\right) = \sum_{\ell=0}^{n} \mathbf{B}_{s,n-\ell} \boldsymbol{\Sigma}_{ss} \mathbf{B}'_{s,n-\ell} + \sum_{\ell=0}^{n} \mathbf{B}_{d,n-\ell} \boldsymbol{\Sigma}_{dd} \mathbf{B}'_{d,n-\ell} + \sum_{\ell=0}^{n} \mathbf{B}_{m,n-\ell} \boldsymbol{\Sigma}_{mm} \mathbf{B}'_{m,n-\ell} + \sum_{\ell=0}^{n} \mathbf{B}_{e,n-\ell} \boldsymbol{\Sigma}_{ee} \mathbf{B}'_{e,n-\ell} + \sum_{\ell=0}^{n} \mathbf{B}_{s,n-\ell} \boldsymbol{\Sigma}_{se} \mathbf{B}'_{e,n-\ell} + \sum_{\ell=0}^{n} \mathbf{B}_{e,n-\ell} \boldsymbol{\Sigma}_{es} \mathbf{B}'_{s,n-\ell} + \sum_{\ell=0}^{n} \mathbf{B}_{d,n-\ell} \boldsymbol{\Sigma}_{de} \mathbf{B}'_{e,n-\ell} + \sum_{\ell=0}^{n} \mathbf{B}_{e,n-\ell} \boldsymbol{\Sigma}_{ed} \mathbf{B}'_{d,n-\ell} + \sum_{\ell=0}^{n} \mathbf{B}_{m,n-\ell} \boldsymbol{\Sigma}_{me} \mathbf{B}'_{e,n-\ell} + \sum_{\ell=0}^{n} \mathbf{B}_{e,n-\ell} \boldsymbol{\Sigma}_{em} \mathbf{B}'_{m,n-\ell}.$$

The first four terms give the contributions to the variance from each of the four shocks; the following six terms arise from the covariances between the exchange rate shocks and the three structural shocks. Using the above FEVD, one can then estimate the importance of supply shocks, demand shocks or monetary policy shocks in the world economy for the explanations of output growth, inflation, interest rates and real effective exchange rates, either for individual variables or any given linear combinations of the variables. These proportions will not add up to unity, due to the error spillover effects between the real effective exchange rates and the three structural shocks. But as we shall show below, due to the relatively small magnitudes of the covariance terms between the real exchange rates and the structural shocks, the proportion of forecast error variances explained by variances of the four shocks add to a number which is very close to unity.

4 Deviations from steady states

So far we have assumed that the deviations from the steady states are given and are covariance stationary, as required by the NK model. In practice, however, such deviations must be identified and measured consistently. In cases where the variables under consideration are either stationary or trend-stationary, the steady state values are either fixed constants or can be approximated by linear trends, and the deviations in the NK model can be replaced by realised values with constant terms or linear trends added to the equations (as appropriate) to take account of the nonzero deterministic means of the stationary or trend stationary processes. But there exists ample evidence that most macroeconomic variables, including inflation and interest rates, real exchange rate and real output, are likely to contain stochastic trends and could be cointegrated. Common stochastic trends at national and global levels can lead to within country as well as between country cointegration. The presence of such stochastic trends must be appropriately taken into account in the identification and estimation of steady state values (and hence the deviations), otherwise the estimates of the structural parameters and the associated impulse responses can be badly biased even in large samples.

There are a variety of methods that can be used to handle permanent components, some of which have been recently discussed by Fukac and Pagan (2010). Here we follow DPSS and measure the steady states as the long-horizon forecasts from an underlying global vector error correcting model (VECM). We also contrast the results obtained using this approach with the alternative often favoured in the literature where inflation and interest rates and real effective exchange rates are treated as stationary, and the output deviations are computed using the Hodrick-Prescott filter.

The global model is specified in terms of the realised values denoted by $\mathbf{x}_t = (\mathbf{x}'_{0t}, \mathbf{x}'_{1t}, ..., \mathbf{x}'_{Nt})'$, with the deviations given by

$$\widetilde{\mathbf{x}}_t = \mathbf{x}_t - \mathbf{x}_t^P,$$

where \mathbf{x}_t^P denotes the permanent component of \mathbf{x}_t . \mathbf{x}_t^P is further decomposed into deterministic and stochastic components

$$\mathbf{x}_t^P = \mathbf{x}_{d,t}^P + \mathbf{x}_{s,t}^P$$
, and $\mathbf{x}_{d,t}^P = \boldsymbol{\mu} + \mathbf{g}t$,

where $\boldsymbol{\mu}$ and \mathbf{g} are $k \times 1$ vectors of constants and t a deterministic time trend. The steady state (permanent-stochastic component) \mathbf{x}_{st}^{P} , is then defined as the 'long-horizon forecast' (net of the permanent-deterministic component)

$$\mathbf{x}_{s,t}^{P} = \lim_{h \to \infty} E_t \left(\mathbf{x}_{t+h} - \mathbf{x}_{d,t+h}^{P} \right) = \lim_{h \to \infty} E_t \left[\mathbf{x}_{t+h} - \boldsymbol{\mu} - \mathbf{g}(t+h) \right].$$

In the case where \mathbf{x}_t is trend stationary then $\mathbf{x}_{s,t}^P = \mathbf{0}$, and we revert back to the familiar case where deviations are formed as residuals from regressions on linear trends. However, in general, $\mathbf{x}_{s,t}^P$ is nonzero and must be estimated from a multivariate time series model of \mathbf{x}_t that allows for stochastic trends and cointegration. Once a suitable multivariate model is specified, it is then relatively easy to show that $\mathbf{x}_{s,t}^P$ corresponds to a multivariate Beveridge-Nelson (1981) decomposition as argued by Garratt *et al.* (2006). The economic model used to provide the long-horizon forecasts is a global VAR (GVAR) which takes account of unit roots and cointegration in the global economy (within as well as across economies). DPSS provide more detail on the GVAR and explain how it can be regarded as the reduced form of a structural model such as the MCNK considered here.

For each country, i = 0, 1, 2, ..., N, the global VAR model consists of VARX^{*} models of the form:

$$\mathbf{x}_{it} = \mathbf{h}_{i0} + \mathbf{h}_{i1}t + \mathbf{A}_{i1}\mathbf{x}_{i,t-1} + \mathbf{B}_{i2}\mathbf{x}_{i,t-2} + \mathbf{C}_{i0}\mathbf{x}_{it}^* + \mathbf{C}_{i1}\mathbf{x}_{i,t-1}^* + \mathbf{u}_{it}, \ i = 0, 1, .., N,$$

and the associated VECM, with cointegrating restrictions:

$$\Delta \mathbf{x}_{it} = \mathbf{c}_{i0} - \boldsymbol{\alpha}_i \boldsymbol{\beta}_i' [\mathbf{z}_{i,t-1} - \boldsymbol{\gamma}_i(t-1)] + \mathbf{C}_{i0} \Delta \mathbf{x}_{it}^* + \mathbf{G}_i \Delta \mathbf{z}_{i,t-1} + \mathbf{u}_{it}$$

where $\mathbf{z}_{it} = (\mathbf{x}'_{it}, \mathbf{x}^{*'}_{it})'$, $\boldsymbol{\alpha}_i$ is a $k_i \times r_i$ matrix of rank r_i , and $\boldsymbol{\beta}_i$ is a $(k_i + k_i^*) \times r_i$ matrix of rank r_i . This allows for cointegration within \mathbf{x}_{it} and between \mathbf{x}_{it} and \mathbf{x}^*_{it} . Then as with (8) $\mathbf{z}_{it} = \mathbf{W}_i \mathbf{x}_t$, and by the same process as above we can stack the N + 1 individual country models and solve for the GVAR specification

$$\mathbf{x}_t = \mathbf{a}_0 + \mathbf{a}_1 t + \mathbf{F}_1 \mathbf{x}_{t-1} + \mathbf{F}_2 \mathbf{x}_{t-2} + \mathbf{u}_t.$$
(26)

This is a standard VAR specification and can be readily used to derive \mathbf{x}_t^P as the long-horizon forecasts of \mathbf{x}_t .

Various GVARs have been widely used for a variety of purposes.⁹ The version used to calculate the long-horizon forecasts is estimated over the same sample, 1979Q4-2006Q4, for the same 33 countries, explaining the same variables (output, inflation, short interest rates, and exchange rates), with the addition of the price of oil, which is included as an endogenous variable in the US VARX^{*} model. These 131 endogenous variables are driven by 82 stochastic trends and 49 cointegrating relations. Weak exogeneity of the foreign variables for the individual VARX^{*} equations is rejected only in 8.4% of the cases at the 5% level. Other versions of the GVAR include financial variables, but these have been excluded for comparability with MCNK model which does not include them.¹⁰

The long-horizon forecasts from this GVAR model provide estimates of the steady states \mathbf{x}_t^P , which match the economic concept of a steady state and are derived from a multivariate economic rather than a univariate statistical model, so they will reflect the long-run cointegrating relationships and stochastic trends in the system. The deviations from steady states used as variables in the MCNK model, $\tilde{\mathbf{x}}_t = \mathbf{x}_t - \mathbf{x}_t^P$, are uniquely identified and stationary by construction so avoiding the danger of spurious regression.

The measures of steady state depend on the underlying economic model, which seems a desirable property. However, they may be sensitive to misspecification and it is possible that intercept shifts, broken trends or other forms of structural instability not allowed for in the estimated economic model, will be reflected in the measured deviations from steady state. For instance, Perron and Wada (2009) argue that the difference between the univariate BN decomposition and other methods of measuring trend US GDP are the artifacts created by neglect of the change in slope of the trend function in 1973. Although our estimation period is all post 1973, so this is not an issue, and various tests indicate that the estimated GVARs seem structurally stable, possible structural breaks could be dealt with using the average long-horizon forecasts from models estimated over different samples. The evidence in Pesaran, Schuermann and Smith (2009) indicates that averaging over observation windows improves forecasts. The extension of this procedure from forecasting to the estimation of steady states is an area for further research.

⁹See for instance Dees, di Mauro, Pesaran and Smith (2007), Dees, Holly, Pesaran and Smith (2007), Pesaran, Smith and Smith (2007), or Pesaran, Schuermann and Smith (2009).

¹⁰Details of the estimated GVAR are provided in a supplement available from the authors upon request.

5 Estimates and solution

The structural equations are estimated for each country separately using the inequality-constrained instrumental variables method.¹¹ DSGE models are often estimated by Bayesian methods but given the size of the model this would be a demanding task. The parameters are estimated subject to the theory restrictions discussed in Subsection 2.1. Where the constraints are not satisfied, the parameters are set to their boundary values and the choice between any alternative estimates that satisfy the constraints is based on the in-sample prediction errors.¹² As to be expected, there is a considerable degree of heterogeneity in the estimates, with those for Latin American countries often being the outliers.¹³

The estimation sample for all equations starts in t = 1980Q1 and ends in 2006Q3 for the Phillips curve and IS equations (due to the presence of future dated variables), and ends in 2006Q4 for the Taylor rule and exchange rate equations. An exception is the Phillips curve for Argentina which is estimated over the sub-sample 1990Q1-2006Q3. The parameters of the structural equations (PC, IS and Taylor rule) are estimated by the IV method using the following as instruments: a vector of ones, $\tilde{\pi}_{i,t-1}, \tilde{y}_{i,t-1}, \tilde{r}_{i,t-1}, \tilde{r}_{e_{i,t-1}}, \tilde{\pi}_{it}^*, \tilde{y}_{it}^*, \tilde{r}_{it}^*$, and \tilde{p}_t^o , except for Saudi Arabia where $\tilde{r}_{i,t-1}$ and \tilde{r}_{it}^* are excluded as there is no interest series for this country. The European countries belonging to the Economic and Monetary Union (EMU) are here considered separately, but an aggregation of these countries into a single region as in DdPS could also be envisaged. The exchange rate equation is estimated by OLS. Table 1 provides summary statistics for the estimates obtained for all the 33 countries in the global DSGE model, and Table 2 gives detailed estimates for eight major economies.¹⁴ We now comment briefly on the estimates.

5.1 Country-specific parameter estimates

The parameters of the Phillips curve, (2), are estimated subject to the inequality restrictions $\beta_{ib} \geq 0$, $\beta_{if} \geq 0$, $\beta_{ib} + \beta_{if} \leq 0.99$, and $\beta_{iy} \geq 0$. Since under $\beta_{ib} = \beta_{if} = 0$, the third restriction, $\beta_{ib} + \beta_{if} \leq 0.99$, is satisfied, there are 14 possible specifications. All specifications are estimated and from those satisfying the restrictions the one with the lowest in-sample mean squared prediction error is selected. Application of this procedure to Argentina over the full sample resulted in the estimates, $\hat{\beta}_{ib} = \hat{\beta}_{if} = \hat{\beta}_{iy} = 0$, which does not seem plausible and could be due to structural breaks, so the PC for Argentina was estimated over the sub-sample, 1990Q1-2006Q3, which gave the somewhat more plausible estimates of $\beta_{if} = 0.53$, $\hat{\beta}_{ib} = \hat{\beta}_{iy} = 0$. In the case of 7 countries, the

¹¹Inference in inequality constrained estimation is non-standard and will not be addressed here. Gouriéroux et al. (1982) consider the problem in the case of least squares estimation.

¹²Pesaran and Smith (1994, p. 708) discuss the relationship between this criterion and the IV minimand.

¹³There is also evidence of misspecification in a number of the estimated equations, but since we wished to consider a tight specification that corresponds to the standard theory we did not add extra lags or more global variables to reduce the extent of the misspecification or to improve the fit of the regressions.

¹⁴Full details of the country-specific estimates are provided in a supplement which is available from the authors upon request.

IV estimates satisfied all the constraints. Also the coefficient of inflation expectations, β_{if} , turned out to be positive in all cases and is generally much larger than the coefficient of lagged inflation, β_{ib} . The mean value of β_{iy} at 0.11 is very close to the standard prior in the literature, although this average hides a wide range of estimates obtained across countries.

	Mean	# Constrained	UC Mean	Constraint			
Phillips curve - Equation (2), N=33							
β_{ib}	0.12 10		0.17	$\beta_{ib} {\geq} 0$			
β_{if}	0.80	0	0.80	$\beta_{if} {\geq} 0$			
β_{iy}	0.11	7	0.14	$\beta_{iy} \ge 0$			
$\beta_{ib} + \beta_{if}$	0.93	22	0.80	$\beta_{ib}{+}\beta_{if}{\leq}0.99$			
IS curve -	Equatio	n (3), N=33					
α_{ib}	0.27	0	0.27				
α_{ir}	-0.20	18	-0.43	$\alpha_{ir} \leq 0$			
α_{ie}	0.02	0	0.02				
α_{iy*}	0.79	2	0.84	$\alpha_{iy*} \ge 0$			
Taylor Rule - Equation (4), N=32							
γ_{ib}	0.59	0	0.59				
$\gamma_{i\pi}$	0.24	4	0.28	$\gamma_{ir} {\geq} 0$			
γ_{iy}	0.06	11	0.09	$\gamma_{iy} \ge 0$			
Exchange rates - Equation (5) , N=32							
ρ_i	0.67	0	0.67	$ \rho_i < 1$			

Table 1: Distribution of inequality-constrained IV estimates using GVAR estimates of deviations from steady states

Note: The estimation sample begins in t=1980Q1 and ends in 2006Q3 for the PC and IS equations, and 2006Q4 for the Taylor rule and exchange rate equations. An exception is the Phillips curve in Argentina which is estimated over the sub-sample 1990Q1-2006Q3. N is the number of countries for which the equations were estimated. The column headed "Mean" gives the average over all estimates, constrained and unconstrained. The column headed "# Constrained" gives the number of estimates constrained at the boundary. The column headed "UC Mean" gives the mean of the unconstrained estimates. Individual country results are available in a supplement available upon request. Results for selected countries are provided in Table 2.

Initially the IS equation was estimated with expected future output deviations included. The coefficient of the future output variable, α_{if} , was negative in 3 countries (Germany, New Zealand and Saudi Arabia), insignificantly positive in 16 countries and significantly positive in 14. In only 11 countries was the coefficient of the real interest rate, α_{ir} , negative as would be expected from the theory. There seemed to be an association between a significant coefficient on the future output variable and a positive real interest rate coefficient, since in 10 out of the 14 countries where the coefficient of future output was significant, the coefficient on the real interest rate was positive.

Various restricted versions of the equation were considered, including setting $\alpha_{if} = 1$, $\alpha_{ib} + \alpha_{if} = 1$, and $\alpha_{if} = 0$. The specification that imposed $\alpha_{if} = 0$ gave the maximum number of countries with negative real interest rate effects. Given the importance of having a negative interest rate effect in the IS curve for the monetary transmission mechanism, we opted for the IS specification without the future output variable, and estimated the parameters of (3) subject to the constraints $\alpha_{ir} \leq 0$ and $\alpha_{iy*} \geq 0$, following the same procedure as before. The unrestricted equation was chosen for 14 countries. Including \tilde{y}_{it}^* tended to produce a more negative and significant estimate of the interest rate effect and, in the case of the US, the estimate of the interest rate coefficient was negative only when \tilde{y}_{it}^* was included in the IS equation. The estimate of the coefficient of the real exchange rate variable averaged to about zero, but with quite a large range of variations across the different countries.

The Taylor Rule, (4) was estimated subject to the constraints $\gamma_{iy} \geq 0$ and $\gamma_{i\pi} \geq 0$. The unrestricted equation was chosen for 18 countries out of the 32 possible Taylor rule equations. Recall that there is no interest rate equation for Saudi Arabia. In the case of Malaysia a fully constrained specification with $\gamma_{iy} = \gamma_{i\pi} = 0$, resulted, and in 3 other countries we obtained the restricted case with $\gamma_{i\pi} = 0$. In 11 countries, including the US, we ended up with $\gamma_{iy} = 0$.

For the real effective exchange rate equation, (5), the OLS estimates of ρ_i ranged from 0.34 to 0.86, confirming that this is a stable process, as one would expect given that we are using deviations from the steady states.

Table 2 shows the inequality-constrained IV estimates for some of the major economies. There is a very strong output effect in the Chinese Phillips curve, and to a lesser extent in the Japanese and the US Phillips curve (with $\beta_{iy} \ge 0.1$), while this effect is somewhat lower in the other countries. There are strong real interest rate effects in the IS curves for the US and Canada, while the estimates for the European countries are either close to zero or constrained at the boundary. Finally, there are strong foreign output effects in the IS curves for all countries, except for Japan.

Hom steady states for eight major economies								
	US	China	Japan	Germany	France	UK	Italy	Canada
			Phillips	curve - Equ	ution (2)			
β_{ib}	0.22	0.00	0.11	0.10	0.00	0.12	0.38	0.22
β_{if}	0.77	0.86	0.84	0.89	0.99	0.87	0.61	0.77
β_{iy}	0.10	0.14	0.13	0.04	0.08	0.05	0.04	0.02
			IS cu	ırve - Equat	ion (3)			
α_{ib}	0.21	0.72	0.15	0.03	0.00	0.54	0.23	0.37
α_{ir}	-0.98	-0.47	-0.23	0.00	-0.02	0.00	0.00	-1.41
α_{ie}	-0.01	0.02	0.02	-0.04	-0.01	0.25	-0.02	0.12
α_{iy*}	0.74	0.31	0.15	1.10	0.64	0.95	0.73	0.89
	Taylor Rule - Equation (4)							
γ_{ib}	0.79	0.98	0.82	0.62	0.94	0.74	0.82	0.51
$\gamma_{i\pi}$	0.28	0.11	0.21	0.27	0.04	0.20	0.20	0.42
γ_{iy}	0.00	0.00	0.15	0.04	0.03	0.01	0.01	0.00
Exchange rates - Equation (5)								
ρ_i		0.78	0.76	0.54	0.68	0.53	0.73	0.84

Table 2: Inequality-constrained IV estimates using GVAR estimates of deviations from steady states for eight major economies

Note: The estimation sample begins in t=1980Q1 and ends in 2006Q3 for the PC and IS equations, and 2006Q4 for the Taylor rule and exchange rate equations. Individual country results are available in a Supplement available upon request.

5.2 Solution and covariance matrix of the shocks

Details of the method used to solve (14), $\mathbf{B}\Phi^2 - \Phi + \mathbf{A} = \mathbf{0}$, are given in the Appendix. It involves an iterative back-substitution procedure starting with an arbitrary initial choice of Φ , which was set to an identity matrix. As a check against multiple solutions, we also started the iterations with an initial value of Φ that had units along the diagonal and the off diagonal terms were drawn from a uniform distribution over the range -0.5 to +0.5. Both initial values resulted in the same solution.

The multi-country NK model is solved for all time periods in our estimation sample, and allows us to obtain estimates of all the structural shocks in the model. Altogether there are 130 different shocks; 98 structural and 32 reduced form. Denote the shock of type k = s, d, m, e in country i = 1, 2, ..., 33 at time t = 1980Q1 - 2006Q4 by $\varepsilon_{i,k,t}$. It is now possible to compute pairwise correlations of any pair of shocks both within and across countries. In Table 3 we provide averages of pair-wise correlations across the four types of shocks. Just to be clear the average pair-wise correlation of supply shocks is computed by averaging over the $(33 \times 32)/2 = 528$ pairs of correlation coefficients from the 33 supply shocks, and similarly the average pair-wise correlation coefficients of supply and demand shocks is computed by averaging over $(33 \times 34)/2 = 561$ pairs of supply-demand shocks.

	Supply	Demand	Mon. Pol.	Ex. Rate
Supply	0.495	0.166	0.040	0.048
Demand		0.067	0.063	-0.005
Mon. Pol.			0.139	-0.043
Ex. Rate				0.049

Table 3: Average pair-wise correlations of shocks using GVAR deviations.

The largest average correlations are among supply shocks, at 0.495; the other correlations are all less than 0.17. By comparison, the average pair-wise correlations of shocks of different types (given as the off-diagonal elements in Table 3) are small, with the largest figure given by the average correlation of demand and supply shocks given by 0.166. The other average correlations across the different types of shocks are small. This is in line with our maintained identifying assumption that supply, demand and monetary policy shocks are orthogonal.

Consider now the problem of consistent estimation of the covariance matrix of shocks defined by (20). One possibility would be to estimate the non-zero blocks Σ_{kl} , k, l = s, d, m, e with the sample covariance matrix using the estimates of ε_t , defined by (17) and denoted by $\hat{\varepsilon}_t$. For instance, Σ_{ss} can be estimated by $\sum_{t=1}^{T} \hat{\varepsilon}_{st} \hat{\varepsilon}'_{st}/T$. These estimates of the component matrices can then be inserted in (20) to provide an estimate of Σ_{ε}^0 , say $\hat{\Sigma}_{\varepsilon}^0$. However, since the dimension of the endogenous variables, k = 130, is larger than the time series dimension, T = 108, $\hat{\Sigma}_{\varepsilon}^0$ is not guaranteed to be a positive definite matrix. While the estimates of the individual correlations are consistent, the estimate of the whole matrix is not when T < N. This is an important consideration when we come to compute bootstrapped error bands for the impulse response functions. The same issue arises in other contexts including mean-variance portfolio optimisation where the number of assets is large.

A number of solutions have been suggested in the literature. Ledoit and Wolf (2004) consider an estimator which is a convex linear combination of the unrestricted sample covariance matrix and an identity matrix and provide an estimator for the weights. Friedman, Hastie and Tibshirani (2008) apply the lasso penalty to loadings in principal component analysis to achieve a sparse representation. Fan, Fan and Lv (2008) use a factor model to impose sparsity on the covariance matrix. Bickel and Levina (2008) propose thresholding the sample covariance matrix, where the threshold parameter is chosen using cross validation.

The procedure we use, starts from the fact that the diagonal matrix, $diag(\hat{\Sigma}_{\varepsilon}^{0})$, which has $(\hat{\sigma}_{1,ss}^{2},...,\hat{\sigma}_{N+1,ss}^{2},\hat{\sigma}_{1,dd}^{2},...,\hat{\sigma}_{N,ee}^{2})'$ on the diagonal and zeros elsewhere, is certainly positive definite. Thus one can use a convex combination of $\hat{\Sigma}_{\varepsilon}^{0}$ and $diag(\hat{\Sigma}_{\varepsilon}^{0})$, which shrinks the sample covariance matrix towards its diagonal, to obtain a positive definite matrix. Such a simple shrinkage estimator of the covariance matrix is given:

$$\hat{\Sigma}^{0}_{\varepsilon}(\varrho) = (1-\varrho)diag\left(\hat{\Sigma}^{0}_{\varepsilon}\right) + \varrho\hat{\Sigma}^{0}_{\varepsilon}.$$
(27)

We experimented with different values of ρ , and found that $\hat{\Sigma}^0_{\varepsilon}(\rho)$ is positive definite for all values of $\rho \leq 0.4$. Accordingly, the initial estimates of the IRFs and FEVDs are based on the shrinkage covariance matrix, $\hat{\Sigma}^{0}_{\varepsilon}(0.4)$. We then examine the sensitivity of the IRFs to the choice of covariance matrix. Since calculation of generalised IRFs does not require the covariance matrix to be positive definite we can compare the IRFs from the shrinkage covariance matrix, $\hat{\Sigma}^{0}_{\varepsilon}(0.4)$ with the IRFs from $\hat{\Sigma}^{0}_{\varepsilon}$, as well as the diagonal covariance matrix, $diag(\hat{\Sigma}^{0}_{\varepsilon})$, and a block diagonal covariance, $Bdiag(\hat{\Sigma}^{0}_{\varepsilon})$ matrix, which sets the covariances between the exchange rate shocks and the structural shocks in $\hat{\Sigma}^{0}_{\varepsilon}$, to zero.

6 Analysis of shocks

A large number of possible counter-factual scenarios can be considered differing in the type of shock, the target country, the nature of the error spillover effects, and the structure of the equations in the global model. We consider the time profiles of the effects of a US monetary policy shock, and global supply and demand shocks on output, inflation and interest rates across the 33 countries. In this section we use the shrinkage covariance matrix estimator defined by (27) with $\rho = 0.40$. This assumes a bordered covariance matrix, with non-zero covariances between structural shocks of the same type, and with unrestricted covariances between the structural shocks and exchange rate shocks.

Notice that we are measuring the effects of an unexpected one period shock not on the variables, but on their deviations from steady states. To examine the effects of shocks on the variables themselves, we would also need to consider the changes in their steady states. The system is stable and following these shocks the variables converge to their steady state values within 5 to 6 years in the vast majority of cases. Although there are only short lags in the system, no more than one period, and strongly forward looking behaviour in the Phillips curve, there is complicated dynamics and some slow adjustment to shocks. The largest eigenvalue of the system is 0.975. Many of the eigenvalues are complex, so adjustments often cycle back to zero. Inflation is a forward-looking variable in this model, so it jumps as expectations adjust to a shock, while interest rates respond strongly to inflation.

6.1 US monetary policy shock

We first consider a contractionary US monetary policy shock, $\mathbf{a}'_{m} \boldsymbol{\varepsilon}^{0}_{t}$, where \mathbf{a}_{m} has zeros except for the element corresponding to, $\varepsilon_{0,mt}$, which is set to unity. Given that this shock has been widely considered in the literature, it is worth simulating it with the MCNK model for comparison purposes. The US monetary policy shock raises the US interest rate on impact by one standard error (around 22 basis points per quarter), which also simultaneously impacts interest rates in other countries through the contemporaneous dependence of monetary policy shocks as captured by the off diagonal elements of $\hat{\Sigma}_{mm}$.

Figures 1a-1c show the effect of a contractionary US monetary shock on interest rates, inflation and output for 26 countries. The results for the five Latin American countries (Argentina, Brazil, Chile, Mexico, Peru), Indonesia and Turkey are excluded as they tend to be outliers due to the much higher levels of inflation and nominal interest rates experienced in these economies over our estimation sample.¹⁵ Also to focus on the differences across countries, the graphs only show the point estimates. Bootstrapped confidence bounds will be considered below.

The monetary policy shock raises interest rates in the US by one standard error, 22 basis points, and interest rates rise almost everywhere else. The mean change for other countries amounts to an increase of 6 basis points, though this is skewed by Argentina, not shown on the graph, and the median is 2 basis points. Interest rates then move below their steady state values very quickly to offset the shock and by quarter 4 they are lower almost everywhere, by -15 basis points in the US; for the other countries the mean is -16 basis points, the median -10 basis points, with the mean skewed to the left by Chile, not shown on the graph. The effect of the monetary policy shock on interest rates in other countries is of the same order of magnitude as in the US. All the interest rates are close to their steady state values within five years, except for the interest rates in Norway which take longer to settle down.

The US monetary policy shock depresses inflation and output, which is consistent with the standard results, e.g. Kim (2001), and output and inflation return to close to steady state within five years for inflation, and six years for output. By quarter 4, US inflation is -0.18 per cent and US output -0.50 per cent below their steady state values. The reduction in US inflation and output in response to the monetary policy shock has a similar shape to that of Smets and Wouters (2007, Fig. 6). The major difference is that whereas in their model a monetary policy shock causes interest rates to go up then slowly return to zero, in our model the monetary policy shock initially raises interest rates, but this is quickly offset by the effects of the relatively sharp falls in output and inflation. This rapid stabilising response occurs despite the fact that there is quite a lot of inertia in our Taylor rules, which have a coefficient of lagged interest rate that averages 0.59. The effects on inflation and output in other countries are similar to those in the US. On average after four quarters, inflation in countries other than the US is lower by -0.18 per cent (per quarter), the same as the US, and output is lower by -0.64 per cent, rather more than the US. The US variables tend to return to their steady state values relatively quickly compared to other countries. The results show that a US monetary policy shock has a rather large global impact in this model.

¹⁵The excluded countries show the same qualitative patterns in their impulse response functions.

Figure 1a: Impulse responses of a one standard error US monetary policy shock on interest rates (per cent per quarter)



Figure 1b: Impulse responses of a one standard error US monetary policy shock on inflation (per cent per quarter)



Figure 1c: Impulse responses of a one standard error US monetary policy shock on output (per cent per quarter)



6.2 Global supply and demand shocks

We now consider a global inflationary supply shock, $\mathbf{a}'_{s} \boldsymbol{\varepsilon}^{0}_{t}$, where the non-zero elements of \mathbf{a}_{s} are PPP GDP weights (that add up to one), associated with the N + 1 supply shocks, $\boldsymbol{\varepsilon}_{i,st}$, in $\boldsymbol{\varepsilon}^{0}_{t}$. Figures 2a-2c show the effects of a unit (one standard error) global supply shock on inflation, output and interest rates across the 26 countries as in Figures 1a-1c. The supply shock causes inflation and interest rates to increase on impact, but then they both fall below their steady state values relatively rapidly, before slowly returning back to the steady states. The global supply shock has quite a large impact. In the US the supply shock increases inflation by 1.4 per cent, which is then reversed to -1.4 per cent after two quarters before returning to its steady state value. The pattern is similar across other countries, though the impact effect on the US is rather higher than the average increase in inflation experienced in other countries, which is 1.0 per cent rather than 1.4 per cent in the US. Similarly, the reduction in US inflation after two quarters is rather larger than the average fall in inflation in other countries. Figure 2a: Impulse responses of a one standard error global supply shock on inflation (per cent per quarter)



Figure 2b: Impulse responses of a one standard error global supply shock on output (per cent per quarter)



Figure 2c: Impulse responses of a one standard error global supply shock on interest rates (per cent per quarter)



The global supply shock also reduces output across the board with an average effect of -2.4 per cent after 4 quarters. The pattern of dynamic adjustments to the global supply shock is different from the standard closed economy models because cross-variable feedbacks seem to operate at a faster pace: inflation and interest rates rapidly move to offset the effects of the inflationary pressure resulting from the global supply shock.

The effects of a global demand shock (constructed similarly to the global supply shock using PPP GDP weights) on output, inflation and interest rates are summarised in Figures 3a-3c. As expected the demand shock has a positive effect on output, inflation and interest rates. In accord with the theory, in the MCNK model a global demand shock increases output and inflation, while a supply shock reduces output and increases inflation. The global demand shock causes output and interest rates to rise before cycling back to their steady state values. The initial expansionary phase of the shock is relatively long lived and takes around 11 to 15 quarters. The effects of the demand shock across countries are qualitatively similar, but differ markedly in the size of the effects. The effect on US output of 3.7 per cent is in the middle of the cross country distribution of the effects, with the US output returning to its steady state value relatively fast. In some other countries output increases further after impact before returning to steady state. The positive effect on inflation lasts a somewhat shorter period than on output. The shape of the responses by output, inflation and interest rates in the US, to the demand shock are qualitatively similar to those reported by Smets and Wouters (2007, Fig. 2).

Figure 3a: Impulse responses of a one standard error global demand shock on output (per cent per quarter)



Figure 3b: Impulse responses of a one standard error global demand shock on inflation (per cent per quarter)



Figure 3c: Impulse responses of a one standard error global demand shock on interest rates (per cent per quarter)



6.3 Forecast error variance decomposition

Figures 4a-4b show the FEVDs for selected economies. The euro area estimates are obtained by averaging over the FEVDs of member countries using PPP GDP weights. The FEVDs across the different variables add up close to unity, being a little below on impact and a little above after 12 quarters on average. While there are differences across countries, in all cases supply and demand shocks account for most of the variations in output, inflation and interest rate in the long-run, with monetary policy shocks and exchange rate shocks account for more of the variations, around 8-10% each. Monetary policy shocks account for more of the variation in interest rates in Canada than in other countries, though even here it is not a large proportion. On impact supply shocks account for nearly all the variation of inflation, but this drops rapidly and these shocks only account for about half of the variation of inflation in the long-run. Demand shocks account for most of the variations in output on impact, but again this figure drops quite rapidly. Smets and Wouters (2007) also find that monetary policy shocks account for relatively little of the variations in output and inflation in the US.



Figure 4a: Forecast error variance decomposition of the shocks in explaining inflation, output and interest rates for the US, the euro area and China

Note: Q0 refers to the values on impact.

Supply Demand □MP REER Japan Japan Japan Inflation Output Interest Rate 1.2 1.2 1.2 1 1 0.8 0.8 0.8 0.6 0.6 0.6 0.4 0.4 0.4 0.2 0.2 0.2 0 0 0 Q0 Q2 Q4 Q6 Q8 Q10 Q12 Q0 Q2 Q4 Q6 Q8 Q10 Q12 Q0 Q2 Q4 Q6 Q8 Q10 Q12 UK UK UK Interest Rate Output Inflation 1.2 1.2 1.2 1 1 1 0.8 0.8 0.8 0.6 0.6 0.6 0.4 0.4 0.4 0.2 0.2 0.2 0 0 Q0 Q2 Q4 Q6 Q8 Q10 Q12 Q0 Q2 Q4 Q6 Q8 Q10 Q12 Q0 Q2 Q4 Q6 Q8 Q10 Q12 Canada Canada Canada Interest Rate Inflation Output 1.2 1.2 1.2 1 1 1 0.8 0.8 0.8 0.6 0.6 0.6 0.4 0.4 0.4 0.2 0.2 0.2 0 0 0 Q0 Q2 Q4 Q6 Q8 Q10 Q12 Q0 Q2 Q4 Q6 Q8 Q10 Q12 Q0 Q2 Q4 Q6 Q8 Q10 Q12

Figure 4b: Forecast error variance decomposition of the shocks in explaining inflation, output and interest rates for Japan, the UK and Canada

Note: Q0 refers to the values on impact.

6.4 Bootstrapped error bands

As noted earlier we also used a bootstrap procedure, set out in detail in the Appendix, to compute 90% error bands for the impulse responses. The results for the effects of US monetary policy shock and global demand and supply shocks on the US and euro area interest rates, output, and inflation are displayed in Figures 5a-5c. As above, the euro area impulse responses are obtained by averaging over the impulse responses of member countries using PPP GDP weights. These figures show the median (which is almost identical to the mean except for India, not shown) and the 5%

and 95% quantiles of the bootstrap distribution. The results indicate that the effects of the shocks are statistically significant in the sense that the 90% bootstrap bands do not always cover zero. Results for other IRFs are similar and available on request.

Figure 5a: Impulse responses of a one standard error US monetary policy shock on US and euro area interest rates, inflation and output (per cent per quarter, bootstrap median estimates together with 90% bootstrap bands)



Figure 5b: Impulse responses of a one standard error global supply shock on US and euro area inflation, output and interest rates (per cent per quarter, bootstrap median estimates together with 90% bootstrap bands)



Figure 5c: Impulse responses of a one standard error global demand shock on US and euro area output, inflation and interest rates (per cent per quarter, bootstrap median estimates together with 90% bootstrap bands)



7 Alternative specifications

In this paper we have deviated from the empirical NK DGSE modelling literature in two important respects. First, we have estimated the steady states as long horizon expectations using an error correcting GVAR specification, as compared to using a purely statistical de-trending procedure. Second, we have allowed for international linkages across shocks and economies using a full multicountry NK model. In what follows we evaluate the importance of these innovations for our results. We also consider the sensitivity of the impulse responses to alternative specifications of the covariances of the structural shocks.

7.1 Measurement of steady states

As an alternative measure of steady states we considered the familiar Hodrick-Prescott (HP) filter for real output and following the literature assumed that the other variables, namely inflation, interest rate and the real exchange rate are stationary, and thus their steady states can be viewed as constants. We computed the HP filter of log real output using the smoothing parameter of 1600 for all countries. The output deviations based on the HP filter were then computed, which we denote by \tilde{y}_{it}^{HP} , for i = 0, 1, ..., N. The country-specific NK models were then estimated by the IV procedure subject to the same theoretical restrictions as above, with an intercept included to allow for the assumed constant steady state values of the other three variables. The instruments used were an intercept, the lagged values of the country specific endogenous variables, $\tilde{y}_{i,t-1}^{HP}$, $\pi_{i,t-1}$, $r_{i,t-1}$, $r_{i,t-1}$, $r_{i,t-1}$, the current values of the foreign variables \tilde{y}_{it}^{HP*} , π_{it}^* , r_{it}^* , r_{it}^* , r_{it}^* , r_{it}^* , r_{it} and the first difference of the oil price variable, Δp_{it}^o . The results are summarised in Table 4 which has the same format as Table 1.

Table 4: Inequality-constrained IV estimates using HP filtered output deviations with smoothing parameter $\lambda = 1600$ and constant steady states for other variables

	Mean	# Constrained	UC Mean	Constraint			
Phillips curve - Equation (2) , N=33							
β_{ib}	0.16 6		0.20	$\beta_{ib} {\geq} 0$			
β_{if}	0.81	0	0.81	$\beta_{if} \ge 0$			
β_{iy}	0.02	12	0.03	$\beta_{iy} \ge 0$			
$\beta_{ib} + \beta_{if}$	0.97	25	0.92	$\beta_{ib}{+}\beta_{if}{\leq}0.99$			
IS curve -	Equatio	on (3), N=33					
α_{ib}	0.67	0	0.67				
. α_{ir}	-0.19	11	-0.28	$\alpha_{ir} \le 0$			
α_{ie}	0.00 0		0.00				
α_{iy*}	0.43	6	0.53	$\alpha_{iy*} \ge 0$			
Taylor Rule - Equation (4), N=32							
γ_{ib}	0.80	0	0.80				
$\gamma_{i\pi}$	0.18	1	0.18	$\gamma_{ir} \ge 0$			
γ_{iy}	0.07	4	0.08	$\gamma_{iy} \ge 0$			
Exchange rates - Equation (5) , N=32							
ρ_i	0.95	0	0.95	$ \rho_i < 1$			

See the notes to Table 1.

¹⁶The foreign output variable based on HP steady state values are computed as $\tilde{y}_{it}^{*HP} = \sum_{j=0}^{N} w_{ij} \tilde{y}_{jt}^{HP}$.

The estimates in Table 4 are more backward looking than those obtained using GVAR deviations, with slower adjustments and near unit root autocorrelation coefficients for the real effective exchange rates. The effect of output deviations in the Phillips curve is smaller using the HP filter as compared to using the GVAR measures of the steady states - also documented in DPSS. In the IS curve, in addition to larger estimated coefficients for the lagged variables and thus slower adjustments, domestic output deviations are less responsive to foreign output deviations when using \tilde{y}_{it}^{HP} . This significantly reduces an important channel for the international transmission of shocks. The Taylor rule is less responsive to inflation and also adjusts more slowly, the coefficient on the lagged interest rate is 0.8 as compared to 0.6 when using the GVAR deviations.

The patterns of average correlations among the estimated shocks, shown in Table 5, is similar to those using GVAR deviations, with the highest average correlation being between supply shocks.

	Supply	Demand	Mon. Pol.	Ex. Rate
Supply	0.475	0.173	0.049	0.003
Demand		0.089	0.017	-0.006
Mon. Pol.			0.070	0.001
Ex. Rate				-0.002

Table 5: Average correlations among the estimated shocks using HP measure ofsteady state for output and constants for other variables.

Again we used the simple shrinkage estimator of the covariance matrix given by (27) with $\rho = 0.4$. The effect of the slower adjustment implied by the parameters using HP deviations for output and constant steady states for the other variables can be seen in Figure 6, which shows the response of output in the various countries to a contractionary US monetary policy shock. While output declines as in Figure 1a, the return to equilibrium is much slower using HP deviations than using GVAR deviations, probably because of the slower estimated adjustment. In fact many of the impulse responses fail to converge to their steady state values even after 40 quarters, which could be indicative of the possible non-stationary nature of some of the variables included in the country specific DSGE models. A comparison of the results in Figures 1a and 6 clearly show the importance of the de-trending procedure for the multi-country analysis of shocks and their transmission in the global economy. Similar results are also obtained when the impulse responses of global supply and demand shocks are compared across the two approaches to the identification and estimation of steady states.

Figure 6: Impulse responses of a one standard error US monetary policy shock on output using HP deviations and constant steady states (per cent per quarter)



7.2 Shutting off direct international linkages

To evaluate the importance of allowing for direct international linkages, we estimated the MCNK model (with GVAR deviations) under alternative restrictions on the coefficient of the foreign output variable in the IS curve. Initially, only the coefficient of foreign output in the US was set to zero. This caused the unrestricted estimate of the interest rate coefficient to be positive, and led to a restricted IS curve for the US without an interest rate variable, thus cutting off the main transmission route for the operation of the US monetary policy. We also experimented with dropping the foreign variables from all the IS equations. Not surprisingly, this caused the average pair-wise correlation coefficient across the demand shocks to increase from 0.166, reported in Table 3, to 0.229, thus shifting the burden of the international transmission of shocks to the indirect effects as captured by error spillover effects. The impulse response functions also became much less sensible as shown in Figures 7a-7c. In response to a US monetary policy shock, interest rates rise almost everywhere, but the response of output and inflation is much more dispersed as compared to the results from the baseline model.

Figure 7a: Impulse responses of a one standard error US monetary policy shock on interest rates in model without foreign output (per cent per quarter)



Figure 7b: Impulse responses of a one standard error US monetary policy shock on inflation in model without foreign output (per cent per quarter)



Figure 7c: Impulse responses of a one standard error US monetary policy shock on output in model without foreign output (per cent per quarter)



7.3 Choice of error covariance matrices

The impulse responses reported so far are based on the shrinkage covariance matrix, $\hat{\Sigma}^0_{\varepsilon}(0.4)$, which uses a weighted average of the sample moment estimate of (20), $\hat{\Sigma}^0_{\varepsilon}$, with its diagonal, $diag(\hat{\Sigma}^0_{\varepsilon})$. Since the choice of the weight, ϱ , is to some extent arbitrary we thought it is important to investigate the sensitivity of our results to the choice of Σ^0_{ε} and how it is estimated. Accordingly, here we consider four alternative estimates of the error covariance matrix: (a) the sample moment estimate, $\hat{\Sigma}^0_{\varepsilon}$ (b) the diagonal matrix, $diag(\hat{\Sigma}^0_{\varepsilon})$, which cuts off all correlations between shocks, and (c) a block diagonal covariance matrix, $Bdiag(\hat{\Sigma}^0_{\varepsilon})$ which imposes zero covariances between exchange rate and other shocks, but allows each type of shock to be correlated within a block, in addition to (d) the shrinkage estimator used above. Setting the covariances of exchange rates with the other shocks to zero, means that the shocks have no effects on exchange rates since there is no direct feedback from the other variables on the real effective exchange rates.

Figure 8 presents IRFs for the effect of a US monetary policy shock on interest rates, inflation and output, using the four estimates of the error covariance matrices. This is the scenario where sensitivity to the choice of the error covariance matrix seems to be the greatest. The different IRFs show the same qualitative pattern, but there seems to be a consistent ranking of the size of the responses, with $\hat{\Sigma}^0_{\varepsilon}$ yielding the smallest effects followed by $\hat{\Sigma}^0_{\varepsilon}(0.4)$, and $diag(\hat{\Sigma}^0_{\varepsilon})$, with $Bdiag(\hat{\Sigma}^0_{\varepsilon})$ producing the largest effects. Using the block diagonal covariance matrix, $Bdiag(\hat{\Sigma}^0_{\varepsilon})$, where the exchange rate covariances are set to zero amplifies the responses relative both to the sample moment estimate, $\hat{\Sigma}^0_{\varepsilon}$, which allows covariances between the exchange rate and other shocks and the fully diagonal covariance matrix, $diag(\hat{\Sigma}^0_{\varepsilon})$, where all covariances are set to zero.

The results for the fully diagonal covariance matrix are interesting, because they show the effect of shutting off all international transmissions through the error spillover effects. With cross error correlations set to zero, there is no indirect instantaneous transmission of the US monetary policy shock to the euro area, so the effect on euro area interest rates on impact is zero. However, direct transmissions through the equations drives euro area interest rates down quite rapidly as the reduction in US output and inflation, reduces euro area output and inflation, prompting an interest rate response. The difference between the block diagonal and the diagonal versions of Σ_{ε}^{0} suggest that the correlations within each of the three blocks of structural shocks amplify the direct responses, while a comparison of the diagonal and block-diagonal choices with $\hat{\Sigma}_{\varepsilon}^{0}$ indicates that allowing for non-zero correlations between the errors of the exchange rate equations and the three structural shocks have important moderating influences on the global interactions.

Figure 8: Impulse responses of a one standard error US monetary policy shock on US and euro area interest rates, inflation and output under alternative covariance matrices (per cent per quarter)



8 Conclusion

This paper shows that it is possible to estimate, solve and simulate a forward-looking multi-country New Keynesian model and use it to estimate the effects of identified supply, demand and monetary policy shocks. In constructing such a model it is necessary to be cautious with regard to the assumptions made about exchange rates, particularly the treatment of the numeraire, and the patterns of cross country error spillover effects. To obtain theory consistent results, it is also important that a priori sign and stability restrictions predicted by the theory are imposed on the parameters of the country-specific models. For all the economies considered, the qualitative effects of demand and supply shocks are as predicted by the theory. Monetary policy shocks are offset more quickly than is typically obtained in the literature. Global supply and demand shocks are the most important drivers of output, inflation and interest rates in the long run. By contrast monetary or exchange rate shocks have only a short-run role in the evolution of the world economy. Despite the uniformity of the specifications assumed across countries, there are major differences between countries in the size of the effects of the shocks. Changing the degree of international transmission, through the use of alternative covariance matrices and foreign variables in equations changed the estimated size of the effects of the shocks. The results indicate the importance of international connections, directly as well as indirectly through error spillover effects: a US monetary policy shock has effects on output and inflation in other countries that are of the same order of magnitude as its effects on the US. Ignoring global inter-connections as country-specific models do, could lead to misleading conclusions.

It is striking that using GVAR deviations and allowing for direct foreign linkages in the countryspecific models produces sensible results. Either using HP deviations or cutting off direct linkages causes the results to become much less sensible. With HP deviations, adjustment is much slower and it is not clear that all the variables converge back to steady state after a shock. Without direct foreign linkages, the effects of the shocks do not match the theory.

The primary objective of the current paper has been to propose a multi-country version of the standard NK model which allows the identification of the usual types of shocks and provides a framework within which a range of questions about international transmission of shocks can be investigated empirically. There are a number of possible routes for further developments. There may be scope to allow for more global variables in the structural equations, which may reduce the crosscountry correlations and allow the identification of country-specific idiosyncratic shocks. There may be advantages in including financial variables like real equity prices and long term interest rates. Less structural models, like the GVAR of DdPS indicate the importance of the international transmission of financial shocks. There is currently considerable macro-finance research to extend DSGE models to include explanations of the term premium in interest rates, the equity premium, the role of banks, government budget deficit, and the role of foreign assets, which is particularly important given the role of the US dollar as an international store of value, not just a unit of account. Another possible development is to introduce international trade variables, such as exports and imports directly rather than indirectly, as is done in this model through including the real effective exchange rate and a country-specific measure of foreign world output in the IS equation. The proposed multi-country model provides a theoretically coherent and empirically viable framework for such extensions.

Appendix

A.1 Solving and Bootstrapping the MCNK model

A.1.1 Solution

Starting with the canonical representation of the global model given by (13)

$$\boldsymbol{\chi}_t = \mathbf{A}\boldsymbol{\chi}_{t-1} + \mathbf{B}\boldsymbol{E}_{t-1}(\boldsymbol{\chi}_{t+1}) + \boldsymbol{\eta}_t,$$

has the following solution

$$\boldsymbol{\chi}_t = \boldsymbol{\Phi} \boldsymbol{\chi}_{t-1} + \boldsymbol{\eta}_t,$$

where Φ satisfies the quadratic matrix equation

$$\mathbf{B}\boldsymbol{\Phi}^2 - \boldsymbol{\Phi} + \mathbf{A} = \mathbf{0}. \tag{A.1}$$

Solving the quadratic equation, we therefore obtain the reduced form solution in terms of $\tilde{\mathbf{x}}_t$ and the structural shocks, $\boldsymbol{\varepsilon}_t$, as

$$\widetilde{\mathbf{x}}_{t} = \mathbf{\Phi}_{11}\widetilde{\mathbf{x}}_{t-1} + \mathbf{\Phi}_{12}\widetilde{\mathbf{x}}_{t-2} + \mathbf{H}_{0}^{-1}\boldsymbol{\varepsilon}_{t},$$

where Φ_{11} and Φ_{12} are defined by

$$oldsymbol{\Phi} = \left(egin{array}{cc} oldsymbol{\Phi}_{11} & oldsymbol{\Phi}_{12} \ oldsymbol{I}_k & oldsymbol{0} \end{array}
ight).$$

To solve (A.1) for $\mathbf{\Phi}$, we employ a back-substitution procedure which involves iterating on an initial arbitrary choice of $\mathbf{\Phi}$ and $\mathbf{\Psi}$, say $\mathbf{\Phi}_0$ and $\mathbf{\Psi}_0$, and use the recursive relations

$$\mathbf{\Phi}_r = \left(\mathbf{I}_k - \mathbf{B}\mathbf{\Phi}_{r-1}\right)^{-1}\mathbf{A},\tag{A.2}$$

$$\Psi_r = \left(\mathbf{I}_k - \mathbf{B}\Phi_{r-1}\right)^{-1}\mathbf{B},\tag{A.3}$$

where Φ_r and Ψ_r are the values of Φ and Ψ , respectively, at the r^{th} iteration (r = 1, 2, ...) and Ψ is the coefficient matrix in the forward equation

$$\mathbf{z}_t = \mathbf{\Psi} E_{t-1}(\mathbf{z}_{t+1}) + \mathbf{v}_t,$$

with

$$egin{aligned} \mathbf{z}_t &= oldsymbol{\chi}_t - oldsymbol{\Phi} oldsymbol{\chi}_{t-1}, \ \mathbf{v}_t &= (\mathbf{I}_k - \mathbf{B} oldsymbol{\Phi})^{-1} oldsymbol{\eta}_t \end{aligned}$$

See Binder and Pesaran (1995, 1997) for further details. Matlab and Gauss code for this procedure is available at http://ideas.repec.org/c/dge/qmrbcd/73.html. This iterative procedure is continued until one of the following convergence criteria is met

$$\|\mathbf{\Phi}_r - \mathbf{\Phi}_{r-1}\|_{\max} \le 10^{-6} \text{ or } \|\mathbf{\Psi}_r - \mathbf{\Psi}_{r-1}\|_{\max} \le 10^{-6}$$

where the max norm of a matrix $\mathbf{A} = \{a_{ij}\}$ is defined as $\|\mathbf{A}\|_{\max} = \max_{i,j} \{|a_{ij}|\}$.

It follows from (16) that

$$\mathbf{u}_t = \tilde{\mathbf{x}}_t - \mathbf{\Phi}_{11} \tilde{\mathbf{x}}_{t-1} - \mathbf{\Phi}_{12} \tilde{\mathbf{x}}_{t-2}$$

and so

$\boldsymbol{\varepsilon}_t = \mathbf{H}_0 \mathbf{u}_t.$

In numerical calculations all unknown parameters are replaced with the restricted IV estimates described in Section 5.

A.1.2 Computation of bootstrap error bands

We generate B bootstrap samples denoted by $\widetilde{\mathbf{\dot{x}}}_t^{(b)},\,b=1,2,...,B$ from the process

$$\tilde{\mathbf{x}}_{t}^{(b)} = \hat{\mathbf{\Phi}}_{11} \tilde{\mathbf{x}}_{t-1}^{(b)} + \hat{\mathbf{\Phi}}_{12} \tilde{\mathbf{x}}_{t-2}^{(b)} + \hat{\mathbf{H}}_{0}^{-1} \hat{\boldsymbol{\varepsilon}}_{t}^{(b)}, \ t = 1, 2, ..., T,$$
(A.4)

by resampling the structural residuals, $\hat{\boldsymbol{\varepsilon}}_t$, and setting $\tilde{\mathbf{x}}_0^{(b)} = \tilde{\mathbf{x}}_0$ and $\tilde{\mathbf{x}}_{-1}^{(b)} = \tilde{\mathbf{x}}_{-1}$, where $\tilde{\mathbf{x}}_0$ and $\tilde{\mathbf{x}}_{-1}$ are the observed initial data vectors that include the US real exchange rate (or equivalently the US price level). Recall that the multi-country rational expectations model is solved in terms of the US price level rather than the US inflation.

We initially orthogonalise the structural shocks, $\hat{\boldsymbol{\varepsilon}}_t$, by using the inverse of the Choleski factor, $\tilde{\mathbf{P}}$, associated with the Choleski decomposition of the shrinkage covariance matrix, $\hat{\boldsymbol{\Sigma}}_{\varepsilon}(0.4)$, defined by (27). This way we obtain the $k \times 1$ orthogonal vector $\hat{\boldsymbol{\upsilon}}_t = \tilde{\mathbf{P}}^{-1} \hat{\boldsymbol{\varepsilon}}_t$ where its j^{th} element $\hat{\boldsymbol{\upsilon}}_{jt}$, j = 1, 2, ..., k, has unit variance. The bootstrap error vector is then obtained as $\boldsymbol{\varepsilon}_t^{(b)} = \tilde{\mathbf{P}} \hat{\boldsymbol{\upsilon}}_t^{(b)}$, where $\hat{\boldsymbol{\upsilon}}_t^{(b)}$ is the $k \times 1$ vector of re-sampled values from $\{\hat{\boldsymbol{\upsilon}}_{jt}\}_{j=1,2,...,k;t=1,2,...,T}$. Prior to any resampling the structural residuals are recentered to ensure that their bootstrap population mean is zero.

Once a set of $\tilde{\mathbf{x}}_{t}^{(b)}$, b = 1, 2, ..., B are generated, US inflation is computed from the US price level so that $\tilde{\mathbf{x}}_{it}^{(b)}$ is constructed, with the corresponding foreign variables, $\tilde{\mathbf{x}}_{it}^{*(b)}$, computed using the trade weights. For each bootstrap replication the individual country models are then estimated by the inequality constrained IV procedure, subject to the constraints given in Section 5, ensuring that any constraint which binds for the estimates based on historical realisations are also imposed on the bootstrap estimates.

The country specific models in terms of $\widetilde{\mathbf{x}}_{it}^{(b)}$ are given by

$$\hat{\mathbf{A}}_{i0}^{(b)} \widetilde{\mathbf{x}}_{it}^{(b)} = \hat{\mathbf{A}}_{i1}^{(b)} \widetilde{\mathbf{x}}_{i,t-1}^{(b)} + \hat{\mathbf{A}}_{i2}^{(b)} E_{t-1} \left(\widetilde{\mathbf{x}}_{i,t+1}^{(b)} \right) + \hat{\mathbf{A}}_{i3}^{(b)} \widetilde{\mathbf{x}}_{it}^{*(b)} + \hat{\mathbf{A}}_{i4}^{(b)} \widetilde{\mathbf{x}}_{i,t-1}^{*(b)} + \boldsymbol{\varepsilon}_{t}^{(b)}$$

and are subsequently combined yielding the MCNK model

$$\tilde{\mathbf{x}}_{t}^{(b)} = \hat{\mathbf{F}}_{1}^{(b)} \tilde{\mathbf{x}}_{t-1}^{(b)} + \hat{\mathbf{F}}_{2}^{(b)} \tilde{\mathbf{x}}_{t-2}^{(b)} + \hat{\mathbf{F}}_{3}^{(b)} E_{t-1} \left(\tilde{\mathbf{x}}_{t+1}^{(b)} \right) + \hat{\mathbf{F}}_{4}^{(b)} E_{t-1} \left(\tilde{\mathbf{x}}_{t}^{(b)} \right) + \mathbf{u}_{t}^{(b)}.$$
(A.5)

Solving the quadratic matrix as described earlier, the reduced form solution of (A.5) follows as

$$\widetilde{\mathbf{x}}_{t}^{(b)} = \mathbf{\hat{\Phi}}_{11}^{(b)} \widetilde{\mathbf{x}}_{t-1}^{(b)} + \mathbf{\hat{\Phi}}_{12}^{(b)} \widetilde{\mathbf{x}}_{t-2}^{(b)} + \mathbf{u}_{t}^{(b)},$$

with

$$\hat{\mathbf{u}}_{t}^{(b)} = \widetilde{\mathbf{x}}_{t}^{(b)} - \hat{\mathbf{\Phi}}_{11}^{(b)} \widetilde{\mathbf{x}}_{t-1}^{(b)} - \hat{\mathbf{\Phi}}_{12}^{(b)} \widetilde{\mathbf{x}}_{t-2}^{(b)}$$

and

$$\hat{\boldsymbol{\varepsilon}}_t^{(b)} = \hat{\mathbf{H}}_0^{(b)} \hat{\mathbf{u}}_t^{(b)}$$

For the first bootstrap replication we begin the iterative back-substitution procedure, using the estimated $\hat{\Phi}$ from the actual data as an initial value to compute (A.4) and (A.4), so that for b = 1, $\Phi_0^{(1)} = \hat{\Phi}$. For each subsequent bootstrap replication, b, the initial value is set to the solution of (A.1) obtained under the preceding replication, b-1, so that $\Phi_0^{(b)} = \hat{\Phi}^{(b-1)}$ and $\Psi_0^{(b)} = (\mathbf{I}_k - \hat{\mathbf{B}}^{(b)} \hat{\Phi}^{(b-1)})^{-1} \hat{\mathbf{B}}^{(b)}$. If for a particular bootstrap replication the iterative back-substitution procedure fails to converge after 500 iterations, the initial values for $\Phi_0^{(b)}$ and $\Psi_0^{(b)}$ are set to the identity matrix.

For each bootstrap replication b = 1, 2, ..., B, having estimated the individual country NK models using the simulated data $\tilde{\mathbf{x}}_{t}^{(b)}$, the MCNK model is reconstructed as described above and the impulse responses are calculated $g^{(b)}(n)$, for n = 0, 1, 2, ... These statistics are then sorted in ascending order, and the $(1 - \alpha)100\%$ confidence interval is calculated by using the $\alpha/2$ and $(1 - \alpha/2)$ quantiles, say $q_{\alpha/2}$ and $q_{(1-\alpha/2)}$, respectively of the bootstrap distribution of g(n).

To compute the upper and lower confidence bounds we use 2000 convergent and stationary bootstrap replications. A convergent replication is defined as one where for the corresponding bootstrap sample, the iterative back-substitution procedure described above (see section A.1.1) converges within 500 iterations, whether the initial values for $\Phi_0^{(b)}$ and $\Psi_0^{(b)}$ are set to the identity matrix or otherwise. Having achieved convergence, a bootstrap replication is checked to make sure that it yields a stationary solution. If any of the above two conditions is violated, a new bootstrap sample is computed. For our bootstrap results we had to carry out a total of 2311 bootstrap replications, of which 311 where due to non-convergence of the iterative back-substitution procedure. No bootstrap replications were found to be non-stationary.

References

Beveridge S, Nelson CR. (1981). A new approach to the decomposition of economic time series into permanent and transitory components with particular attention to the measurement of the 'business cycle'. Journal of Monetary Economics 7: 151-174.

Bickel PJ, Levina E. (2008). Covariance regularization by thresholding. The Annals of Statistics 36: 2577-2604.

Binder M, Pesaran MH. (1995). Multivariate rational expectations models and macroeconomic modelling: A review and some new results. In Handbook of Applied Econometrics Vol. 1, Pesaran MH, Wickens W (eds.): 655-673.

Binder M, Pesaran MH. (1997). Multivariate linear rational expectations models: characterization of the nature of the solutions and their fully recursive computation. Econometric Theory 13: 877-888.

Blanchard O, Quah D. (1989). The dynamic effects of aggregate demand and supply disturbances. American Economic Review 79: 655-673.

Canova F, Ciccarelli M. (2009). Estimating multi-country VAR models. International Economic Review 50: 929-959.

Canova F, Sala L. (2009). Back to square one: Identification issues in DSGE models. Journal of Monetary Economics 56: 431-449.

Carabenciov I, Ermolaev I, Freedman C, Juillard M, Kamenik O, Korshunov D, Laxton D, Laxton J. (2008). A small multi-country projection model with financial-real linkages and oil prices. IMF working paper 08/280.

Carlstrom CT, Fuerst TS, Paustian M. (2009). Monetary policy shocks, Choleski identification, and DNK models. Journal of Monetary Economics 56: 1014-1021.

Chudik A, Pesaran MH. (2010). Econometric Analysis of High Dimensional VARs Featuring a Dominant Unit, unpublished manuscript, Cambridge University.

Clarida R, Galí J, Gertler M. (1999). The science of monetary policy: A new Keynesian perspective. Journal of Economic Literature 37: 1661-1707.

Dees S, di Mauro F, Pesaran MH, Smith LV. (2007). Exploring the international linkages of the euro area: a global VAR analysis. Journal of Applied Econometrics 22: 1-38.

Dees S, Holly S, Pesaran MH, Smith LV. (2007). Long run macroeconomic relations in the global economy. Economics - The Open-Access, Open-Assessment E-Journal, 2007-3.

Dees S, Pesaran MH, Smith LV, Smith RP. (2009). Identification of new Keynesian Phillips curves from a global perspective. Journal of Money Credit and Banking 41: 1481-1502.

De Walque G, Smets F, Wouters R. (2005). An open economy DSGE model linking the euro area and the US economy. Mimeo.

Eichenbaum M, Evans C. (1995). Some empirical evidence of the effect of monetary policy shocks on exchange rates. Quarterly Journal of Economics 110: 975-1009.

Fan J, Fan Y, Lv J. (2008). High dimensional covariance matrix estimation using a factor model. Journal of Econometrics, 147: 186-197.

Friedman J, Hastie T, Tibshirani R. (2008). Sparse inverse covariance estimation with the graphical lasso. Biostatistics 9: 432–441.

Fuhrer JC, and Rudebusch, GD. (2004). Estimating the Euler equation for output, Journal of Monetary Economics, 51, 1133-1153

Fukac M, Pagan A. (2010). Limited information estimation and evaluation of DSGE models. Journal of Applied Econometrics 25: 55-70.

Galí J, Monacelli T. (2005). Monetary policy and exchange rate volatility in a small open economy. Review of Economic Studies 72: 707–734.

Garratt A, Robertson D, Wright S. (2006). Permanent vs. transitory components and economic fundamentals. Journal of Applied Econometrics 21: 521-42.

Gouriéroux C, Holly A, Montford A.(1982). Likelihood ratio test, Wald test and Kuhn-Tucker test in linear models with inequality constraints on the regression parameters. Econometrica 50: 63-80.

Ireland PN. (2004). Technology shocks in the new Keynesian model. Review of Economics and Statistics 86: 923-936.

Kim S. (2001). International transmission of US monetary policy shocks: evidence from VAR's. Journal of Monetary Economics 48: 339-372.

Ledoit O, Wolf M. (2004). A well-conditioned estimator for large-dimensional covariance matrices. Journal of Multivariate Analysis 88: 365-411.

Lubik TA, Schorfheide F. (2007). Do central banks respond to exchange rate movements? A structural investigation. Journal of Monetary Economics 54: 1069-1087.

Perron P, Wada T. (2009). Lets take a break: trends and cycles in US real GDP. Journal of Monetary Economics 56: 749-765.

Pesaran, MH, T Schuermann, Weiner S. (2004). Modelling Regional Interdependencies using a Global Error-correcting Macroeconometric Model, Journal of Business and Economics Statistics 22: 129-162.

Pesaran MH, Schuermann T, Smith LV. (2009). Forecasting economic and financial variables with global VARs. International Journal of Forecasting 25: 642-675.

Pesaran MH, Smith RJ. (1994). A generalized R^2 criterion for regression models estimated by the instrumental variables method. Econometrica 62: 705-710.

Pesaran MH, Smith LV, Smith RP. (2007). What if the UK or Sweden had joined the Euro in 1999? An empirical evaluation using a global VAR. International Journal of Finance and Economics 12: 55-87.

Smets F, Wouters R. (2007). Shocks and frictions in US business cycles: a Bayesian DSGE approach. American Economic Review 97: 586-606.

Uhlig H. (2005). What are the effects of monetary policy on output? Results from an agnostic identification procedure. Journal of Monetary Economics 52: 381-419.

Woodford M. (2003). Interest and prices: Foundations of a theory of monetary policy. Princeton University Press.

CESifo Working Paper Series

for full list see www.cesifo-group.org/wp (address: Poschingerstr. 5, 81679 Munich, Germany, office@cesifo.de)

- 3017 Alessandro Fedele, Paolo M. Panteghini and Sergio Vergalli, Optimal Investment and Financial Strategies under Tax Rate Uncertainty, April 2010
- 3018 Laurence Jacquet, Take it or Leave it: Take-up, Optimal Transfer Programs, and Monitoring, April 2010
- 3019 Wilhelm Kohler and Jens Wrona, Offshoring Tasks, yet Creating Jobs?, April 2010
- 3020 Paul De Grauwe, Top-Down versus Bottom-Up Macroeconomics, April 2010
- 3021 Karl Ove Aarbu, Demand Patterns for Treatment Insurance in Norway, April 2010
- 3022 Toke S. Aidt and Jayasri Dutta, Fiscal Federalism and Electoral Accountability, April 2010
- 3023 Bahram Pesaran and M. Hashem Pesaran, Conditional Volatility and Correlations of Weekly Returns and the VaR Analysis of 2008 Stock Market Crash, April 2010
- 3024 Stefan Buehler and Dennis L. Gärtner, Making Sense of Non-Binding Retail-Price Recommendations, April 2010
- 3025 Leonid V. Azarnert, Immigration, Fertility, and Human Capital: A Model of Economic Decline of the West, April 2010
- 3026 Christian Bayer and Klaus Wälde, Matching and Saving in Continuous Time: Theory and 3026-A Matching and Saving in Continuous Time: Proofs, April 2010
- 3027 Coen N. Teulings and Nick Zubanov, Is Economic Recovery a Myth? Robust Estimation of Impulse Responses, April 2010
- 3028 Clara Graziano and Annalisa Luporini, Optimal Delegation when the Large Shareholder has Multiple Tasks, April 2010
- 3029 Erik Snowberg and Justin Wolfers, Explaining the Favorite-Longshot Bias: Is it Risk-Love or Misperceptions?, April 2010
- 3030 Doina Radulescu, The Effects of a Bonus Tax on Manager Compensation and Welfare, April 2010
- 3031 Helmut Lütkepohl, Forecasting Nonlinear Aggregates and Aggregates with Timevarying Weights, April 2010
- 3032 Silvia Rocha-Akis and Ronnie Schöb, Welfare Policy in the Presence of Unionised Labour and Internationally Mobile Firms, April 2010

- 3033 Steven Brakman, Robert Inklaar and Charles van Marrewijk, Structural Change in OECD Comparative Advantage, April 2010
- 3034 Dirk Schindler and Guttorm Schjelderup, Multinationals, Minority Ownership and Tax-Efficient Financing Structures, April 2010
- 3035 Christian Lessmann and Gunther Markwardt, Decentralization and Foreign Aid Effectiveness: Do Aid Modality and Federal Design Matter in Poverty Alleviation?, April 2010
- 3036 Eva Deuchert and Conny Wunsch, Evaluating Nationwide Health Interventions when Standard Before-After Doesn't Work: Malawi's ITN Distribution Program, April 2010
- 3037 Eric A. Hanushek and Ludger Woessmann, The Economics of International Differences in Educational Achievement, April 2010
- 3038 Frederick van der Ploeg, Aggressive Oil Extraction and Precautionary Saving: Coping with Volatility, April 2010
- 3039 Ainura Uzagalieva, Evžen Kočenda and Antonio Menezes, Technological Imitation and Innovation in New European Union Markets, April 2010
- 3040 Nicolas Sauter, Jan Walliser and Joachim Winter, Tax Incentives, Bequest Motives, and the Demand for Life Insurance: Evidence from two Natural Experiments in Germany, April 2010
- 3041 Matthias Wrede, Multinational Capital Structure and Tax Competition, April 2010
- 3042 Burkhard Heer and Alfred Maussner, A Note on the Computation of the Equity Premium and the Market Value of Firm Equity, April 2010
- 3043 Kristiina Huttunen, Jukka Pirttilä and Roope Uusitalo, The Employment Effects of Low-Wage Subsidies, May 2010
- 3044 Matthias Kalkuhl and Ottmar Edenhofer, Prices vs. Quantities and the Intertemporal Dynamics of the Climate Rent, May 2010
- 3045 Bruno S. Frey and Lasse Steiner, Pay as you Go: A New Proposal for Museum Pricing, May 2010
- 3046 Henning Bohn and Charles Stuart, Population under a Cap on Greenhouse Gas Emissions, May 2010
- 3047 Balázs Égert and Rafal Kierzenkowski, Exports and Property Prices in France: Are they Connected?, May 2010
- 3048 Thomas Eichner and Thorsten Upmann, Tax-Competition with Involuntary Unemployment, May 2010

- 3049 Taiji Furusawa, Kazumi Hori and Ian Wooton, A Race beyond the Bottom: The Nature of Bidding for a Firm, May 2010
- 3050 Xavier Vives, Competition and Stability in Banking, May 2010
- 3051 Thomas Aronsson and Erkki Koskela, Redistributive Income Taxation under Outsourcing and Foreign Direct Investment, May 2010
- 3052 Michael Melvin and Duncan Shand, Active Currency Investing and Performance Benchmarks, May 2010
- 3053 Sören Blomquist and Laurent Simula, Marginal Deadweight Loss when the Income Tax is Nonlinear, May 2010
- 3054 Lukas Menkhoff, Carol L. Osler and Maik Schmeling, Limit-Order Submission Strategies under Asymmetric Information, May 2010
- 3055 M. Hashem Pesaran and Alexander Chudik, Econometric Analysis of High Dimensional VARs Featuring a Dominant Unit, May 2010
- 3056 Rabah Arezki and Frederick van der Ploeg, Do Natural Resources Depress Income Per Capita?, May 2010
- 3057 Joseph Plasmans and Ruslan Lukach, The Patterns of Inter-firm and Inter-industry Knowledge Flows in the Netherlands, May 2010
- 3058 Jenny E. Ligthart and Sebastian E. V. Werner, Has the Euro Affected the Choice of Invoicing Currency?, May 2010
- 3059 Håkan Selin, Marginal Tax Rates and Tax-Favoured Pension Savings of the Self-Employed – Evidence from Sweden, May 2010
- 3060 Richard Cornes, Roger Hartley and Yuji Tamura, A New Approach to Solving Production-Appropriation Games with Many Heterogeneous Players, May 2010
- 3061 Ronald MacDonald and Flávio Vieira, A Panel Data Investigation of Real Exchange Rate Misalignment and Growth, May 2010
- 3062 Thomas Eichner and Rüdiger Pethig, Efficient Management of Insecure Fossil Fuel Imports through Taxing(!) Domestic Green Energy?, May 2010
- 3063 Vít Bubák, Evžen Kočenda and Filip Žikeš, Volatility Transmission in Emerging European Foreign Exchange Markets, May 2010
- 3064 Leonid V. Azarnert, Après nous le Déluge: Fertility and the Intensity of Struggle against Immigration, May 2010
- 3065 William E. Becker, William H. Greene and John J. Siegfried, Do Undergraduate Majors or Ph.D. Students Affect Faculty Size?, May 2010

3066 Johannes Becker, Strategic Trade Policy through the Tax System, May 2010

- 3067 Omer Biran and Françoise Forges, Core-stable Rings in Auctions with Independent Private Values, May 2010
- 3068 Torben M. Andersen, Why do Scandinavians Work?, May 2010
- 3069 Andrey Launov and Klaus Wälde, Estimating Incentive and Welfare Effects of Non-Stationary Unemployment Benefits, May 2010
- 3070 Simon Gächter, Benedikt Herrmann and Christian Thöni, Culture and Cooperation, June 2010
- 3071 Mehmet Bac and Eren Inci, The Old-Boy Network and the Quality of Entrepreneurs, June 2010
- 3072 Krisztina Molnár and Sergio Santoro, Optimal Monetary Policy when Agents are Learning, June 2010
- 3073 Marcel Boyer and Donatella Porrini, Optimal Liability Sharing and Court Errors: An Exploratory Analysis, June 2010
- 3074 Guglielmo Maria Caporale, Roman Matousek and Chris Stewart, EU Banks Rating Assignments: Is there Heterogeneity between New and Old Member Countries? June 2010
- 3075 Assaf Razin and Efraim Sadka, Fiscal and Migration Competition, June 2010
- 3076 Shafik Hebous, Martin Ruf and Alfons Weichenrieder, The Effects of Taxation on the Location Decision of Multinational Firms: M&A vs. Greenfield Investments, June 2010
- 3077 Alessandro Cigno, How to Deal with Covert Child Labour, and Give Children an Effective Education, in a Poor Developing Country: An Optimal Taxation Problem with Moral Hazard, June 2010
- 3078 Bruno S. Frey and Lasse Steiner, World Heritage List: Does it Make Sense?, June 2010
- 3079 Henning Bohn, The Economic Consequences of Rising U.S. Government Debt: Privileges at Risk, June 2010
- 3080 Rebeca Jiménez-Rodriguez, Amalia Morales-Zumaquero and Balázs Égert, The VARying Effect of Foreign Shocks in Central and Eastern Europe, June 2010
- 3081 Stephane Dees, M. Hashem Pesaran, L. Vanessa Smith and Ron P. Smith, Supply, Demand and Monetary Policy Shocks in a Multi-Country New Keynesian Model, June 2010